AMS/IP Studies in Advanced Mathematics

S.-T. Yau, Series Editor

Heat Kernel and Analysis on Manifolds

Alexander Grigor'yan

American Mathematical Society • International Press

Heat Kernel and Analysis on Manifolds

Studies in Advanced Mathematics

Volume 47

Heat Kernel and Analysis on Manifolds

Alexander Grigor'yan

American Mathematical Society





International Press

•

Shing-Tung Yau, General Editor

2010 Mathematics Subject Classification. Primary 58J35; Secondary 31B05, 35K05, 35P15, 47D07, 53C20.

For additional information and updates on this book, visit www.ams.org/bookpages/amsip-47

Library of Congress Cataloging-in-Publication Data

Grigor'yan, A. (Alexander)
Heat kernel and analysis on manifolds / Alexander Grigor'yan.
p. cm. (AMS/IP studies in advanced mathematics ; v. 47)
Includes bibliographical references and index.
ISBN 978-0-8218-4935-4 (alk. paper)
1. Heat equation. 2. Kernel functions. 3. Riemannian manifolds. 4. Gaussian processes.
I. Title.

QA377.G754 2009 515'.353—dc22

2009034350

Copying and reprinting. Individual readers of this publication, and nonprofit libraries acting for them, are permitted to make fair use of the material, such as to copy a chapter for use in teaching or research. Permission is granted to quote brief passages from this publication in reviews, provided the customary acknowledgment of the source is given.

Republication, systematic copying, or multiple reproduction of any material in this publication is permitted only under license from the American Mathematical Society and International Press. Requests for such permission should be addressed to the Acquisitions Department, American Mathematical Society, 201 Charles Street, Providence, Rhode Island 02904-2294, USA. Requests can also be made by e-mail to reprint-permission@ams.org.

© 2009 by the American Mathematical Society. All rights reserved.

The American Mathematical Society retains all rights

except those granted to the United States Government.

Printed in the United States of America.

© The paper used in this book is acid-free and falls within the guidelines established to ensure permanence and durability. Visit the AMS home page at http://www.ams.org/

Visit the International Press home page at URL: http://www.intlpress.com/

 $10\ 9\ 8\ 7\ 6\ 5\ 4\ 3\ 2\ 1 \qquad 17\ 16\ 15$

To my wife Tatiana

Contents

Preface	xi
Chapter 1. Laplace operator and the heat equation in \mathbb{R}^n	1
1.1. Historical background	1
1.2. The Green formula	2
1.3. The heat equation	4
Notes	13
Chapter 2. Function spaces in \mathbb{R}^n	15
2.1. Spaces C^k and L^p	15
2.2. Convolution and partition of unity	17
2.3. Approximation of integrable functions by smooth ones	20
2.4. Distributions	23
2.5. Approximation of distributions by smooth functions	28
2.6. Weak derivatives and Sobolev spaces	34
2.7. Heat semigroup in \mathbb{R}^n	40
Notes	47
Chapter 3. Laplace operator on a Riemannian manifold	49
3.1. Smooth manifolds	49
3.2. Tangent vectors	53
3.3. Riemannian metric	56
3.4. Riemannian measure	59
3.5. Divergence theorem	64
3.6. Laplace operator and weighted manifolds	67
3.7. Submanifolds	70
3.8. Product manifolds	72
3.9. Polar coordinates in $\mathbb{R}^n, \mathbb{S}^n, \mathbb{H}^n$	74
3.10. Model manifolds	80
3.11. Length of paths and the geodesic distance	85
3.12. Smooth mappings and isometries	91
Notes	95
Chapter 4. Laplace operator and heat equation in $L^{2}(M)$	97
4.1. Distributions and Sobolev spaces	97
4.2. Dirichlet Laplace operator and resolvent	103
4.3. Heat semigroup and L^2 -Cauchy problem	112

CONTENTS

Notes	122
Chapter 5. Weak maximum principle and related topics 5.1. Chain rule in W_0^1 5.2. Chain rule in W^1 5.3. Markovian properties of resolvent and the heat semigroup 5.4. Weak maximum principle 5.5. Resolvent and the heat semigroup in subsets Notes	$123 \\ 123 \\ 127 \\ 130 \\ 135 \\ 143 \\ 149$
Chapter 6. Regularity theory in \mathbb{R}^n 6.1. Embedding theorems 6.2. Two technical lemmas 6.3. Local elliptic regularity 6.4. Local parabolic regularity Notes	151 151 159 162 170 181
 Chapter 7. The heat kernel on a manifold 7.1. Local regularity issues 7.2. Smoothness of the semigroup solutions 7.3. The heat kernel 7.4. Extension of the heat semigroup 7.5. Smoothness of the heat kernel in t, x, y 7.6. Notes 	183 183 190 198 201 208 215
 Chapter 8. Positive solutions 8.1. The minimality of the heat semigroup 8.2. Extension of resolvent 8.3. Strong maximum/minimum principle 8.4. Stochastic completeness Notes 	217 217 219 222 231 241
 Chapter 9. Heat kernel as a fundamental solution 9.1. Fundamental solutions 9.2. Some examples 9.3. Eternal solutions Notes 	243 243 248 259 263
Chapter 10. Spectral properties 10.1. Spectra of operators in Hilbert spaces 10.2. Bottom of the spectrum 10.3. The bottom eigenfunction 10.4. The heat kernel in relatively compact regions 10.5. Minimax principle 10.6. Discrete spectrum and compact embedding theorem 10.7. Positivity of λ_1 10.8. Long time asymptotic of log p_t	265 265 271 275 277 284 287 291 292

viii

NT (000
Notes		293
Chapter 11.1. 11.2. 11.3. 11.4. 11.5. 11.6. Notes	 11. Distance function and completeness The notion of completeness Lipschitz functions Essential self-adjointness Stochastic completeness and the volume growth Parabolic manifolds Spectrum and the distance function 	295 295 296 301 303 313 317 319
Chapter 12.1. 12.2. 12.3. 12.4. 12.5. Notes	12. Gaussian estimates in the integrated form The integrated maximum principle The Davies-Gaffney inequality Upper bounds of higher eigenvalues Semigroup solutions with a harmonic initial function Takeda's inequality	321 321 324 327 331 333 339
Chapter 13.1. 13.2. 13.3. 13.4. 13.5. 13.6. Notes	13. Green function and Green operator The Green operator Superaveraging functions Local Harnack inequality Convergence of sequences of α -harmonic functions The positive spectrum Green function as a fundamental solution	341 341 348 351 355 357 359 362
Chapter 14.1. 14.2. 14.3. 14.4. 14.5. 14.6. 14.7. 14.8. Notes	14. Ultracontractive estimates and eigenvalues Ultracontractivity and heat kernel bounds Faber-Krahn inequalities The Nash inequality The function classes \mathbf{L} and Γ Faber-Krahn implies ultracontractivity Ultracontractivity implies a Faber-Krahn inequality Lower bounds of higher eigenvalues Faber-Krahn inequality on direct products	365 367 368 371 380 381 384 386 388
Chapter 15.1. 15.2. 15.3. 15.4. 15.5. 15.6. Notes	15. Pointwise Gaussian estimates I L^2 -mean value inequality Faber-Krahn inequality in balls The weighted L^2 -norm of heat kernel Faber-Krahn inequality in unions of balls Off-diagonal upper bounds Relative Faber-Krahn inequality and Li-Yau upper bounds	391 391 397 399 402 404 409 414

CONTENTS

 $\mathbf{i}\mathbf{x}$

CONTENTS

Chapter 16. Pointwise Gaussian estimates II	417
16.1. The weighted L^2 -norm of $P_t f$	417
16.2. Gaussian upper bounds of the heat kernel	422
16.3. On-diagonal lower bounds	424
16.4. Epilogue: alternative ways of constructing the heat kernel	428
Notes and further references	429
Appendix A. Reference material	431
A.1. Hilbert spaces	431
A.2. Weak topology	432
A.3. Compact operators	434
A.4. Measure theory and integration	434
A.5. Self-adjoint operators	444
A.6. Gamma function	455
Bibliography	457
Some notation	
Index	477

x

Preface

The development of Mathematics in the past few decades has witnessed an unprecedented rise in the usage of the notion of heat kernel in the diverse and seemingly remote sections of Mathematics. In the paper [217], titled "The ubiquitous heat kernel", Jay Jorgenson and Serge Lang called the heat kernel "... a universal gadget which is a dominant factor practically everywhere in mathematics, also in physics, and has very simple and powerful properties."

Already in a first Analysis course, one sees a special role of the exponential function $t \mapsto e^{at}$. No wonder that a far reaching generalization of the exponential function – the heat semigroup $\{e^{-tA}\}_{t\geq 0}$, where A is a positive definite linear operator, plays similarly an indispensable role in Mathematics and Physics, not the least because it solves the associated heat equation $\dot{u} + Au = 0$. If the operator A acts in a function space then frequently the action of the semigroup e^{-tA} is given by an integral operator, whose kernel is called then the heat kernel of A.

Needless to say that any knowledge of the heat kernel, for example, upper and/or lower estimates, can help in solving various problems related to the operator A and its spectrum, the solutions to the heat equation, as well as to the properties of the underlying space. If in addition the operator A is Markovian, that is, generates a Markov process (for example, this is the case when A is a second order elliptic differential operator), then one can use information about the heat kernel to answer questions concerning the process itself.

This book is devoted to the study of the heat equation and the heat kernel of the Laplace operator on Riemannian manifolds. Over 140 years ago, in 1867, Eugenio Beltrami [29] introduced the Laplace operator for a Riemannian metric, which is also referred to as the Laplace-Beltrami operator. The next key step towards analysis of this operator was made in 1954 by Matthew Gaffney [126], who showed that on geodesically complete manifolds the Laplace operator is essentially self-adjoint in L^2 . Gaffney also proved in [127] the first non-trivial sufficient condition for the stochastic completeness of the heat semigroup, that is, for the preservation of the L^1 -norm by this semigroup. Nearly at the same time S. Minakshisundaram [275] constructed the heat kernel on compact Riemannian manifolds using the parametrix method.

However, it was not until the mid-1970s when the geometric analysis of the Laplace operator and the heat equation was revolutionized in the groundbreaking work of Shing-Tung Yau, which completely reshaped the area. The culmination of this work was the proof by Li and Yau [**258**] in 1986 of the parabolic Harnack inequality and the heat kernel two-sided estimates on complete manifolds of non-negative Ricci curvature, which stimulated further research on heat kernel estimates by many authors. Apart from the general wide influence on geometric analysis, the gradient estimates of Li and Yau motivated Richard Hamilton in his program on Ricci flow that eventually lead to the resolution of the Poincaré conjecture by Grigory Perel'man, which can be viewed as a most spectacular application of heat kernels in geometry¹.

Another direction in heat kernel research was developed by Brian Davies [96] and Nick Varopoulos [353], [355], who used primarily function-analytic methods to relate heat kernel estimates to certain functional inequalities.

The purpose of this book is to provide an accessible for graduate students introduction to the geometric analysis of the Laplace operator and the heat equation, which would bridge the gap between the foundations of the subject and the current research. The book focuses on the following aspects of these notions, which form separate chapters or groups of chapters.

I. Local geometric background. A detailed introduction to Riemannian geometry is given, with emphasis on construction of the Riemannian measure and the Riemannian Laplace operator as an elliptic differential operator of second order, whose coefficients are determined by the Riemannian metric tensor.

II. Spectral-theoretic properties. It is a crucial observation that the Laplace operator can be extended to a self-adjoint operator in L^2 space, which enables one to invoke the spectral theory and functional calculus of self-adjoint operator and, hence, to construct the associated heat semigroup. To treat properly the domains of the self-adjoint Laplacian and that of the associated energy form, one needs the Sobolev function spaces on manifolds. A detailed introduction to the theory of distributions and Sobolev spaces is given in the setting of \mathbb{R}^n and Riemannian manifolds.

III. Markovian properties and maximum principles. The above spectraltheoretic aspect of the Laplace operator exploits its ellipticity and symmetry. The fact that its order is 2 leads to the so-called Markovian properties, that is, to maximum and minimum principles for solutions to the Laplace equation and the heat equation. Various versions of maximum/minimum principles are presented in different parts of the book, in the weak, normal, and strong forms. The Markovian properties are tightly related to the diffusion Markov process associated with the Laplacian, where is reflected in

¹Another striking application of heat kernels is the heat equation approach to the Atiyah-Singer index theorem – see [12], [132], [317].

the terminology. However, we do not treat stochastic processes here, leaving this topic for a prospective second volume.

IV. Smoothness properties. As it is well-known, elliptic and parabolic equations feature an added regularity phenomenon, when the degree of smoothness of solutions is higher than a priori necessary. A detailed account of the local regularity theory in \mathbb{R}^n (and consequently on manifolds) is given for elliptic and parabolic operators with smooth coefficients. This includes the study of the smoothness of solutions in the scale of Sobolev spaces of positive and negative orders, as well as the embedding theorems of Sobolev spaces into C^k . The local estimates of solutions are used, in particular, to prove the existence of the heat kernel on an arbitrary manifold.

V. Global geometric aspects. These are those properties of solutions which depend on the geometry of the manifold in the large, such as the essential self-adjointness of the Laplace operator (that is, the uniqueness of the self-adjoint extension), the stochastic completeness of the heat kernel, the uniqueness in the bounded Cauchy problem for the heat equation, and the quantitative estimates of solutions, in particular, of the heat kernel. A special attention is given to upper bounds of the heat kernel, especially the on-diagonal upper bounds with the long-time dependence, and the Gaussian upper bounds reflecting the long-distance behavior. The lower bounds as well as the related uniform Harnack inequalities and gradient estimates are omitted and will be included in the second volume.

The prerequisites for reading of this books are Analysis in \mathbb{R}^n and the basics of Functional Analysis, including Measure Theory, Hilbert spaces, and Spectral Theorem for self-adjoint operators (the necessary material from Functional Analysis is briefly surveyed in Appendix). The book can be used as a source for a number of graduate lecture courses on the following topics: Riemannian Geometry, Analysis on Manifolds, Sobolev Spaces, Partial Differential Equations, Heat Semigroups, Heat Kernel Estimates, and others. In fact, it grew up from a graduate course "Analysis on Manifolds" that was taught by the author in 1995-2005 at Imperial College London and in 2002, 2005 at Chinese University of Hong Kong.

The book is equipped with over 400 exercises whose level of difficulty ranges from "general nonsense" to quite involved. The exercises extend and illustrate the main text, some of them are used in the main text as lemmas. The detailed solutions of the exercises (about 200 pages) as well as their IATEX sources are available on the web page of the AMS

http://www.ams.org/bookpages/amsip-47

where also additional material on the subject of the book will be posted.

The book has little intersection with the existing monographs on the subject. The above mentioned upper bounds of heat kernels, which were obtained mostly by the author in 1990s, are presented for the first time in a book format. However, the background material is also significantly different from the previous accounts. The main distinctive feature of the foundation

part of this book is a new method of construction of the heat kernel on an arbitrary Riemannian manifold. Since the above mentioned work by Minakshisundaram, the traditional method of constructing the heat kernel was by using the parametrix method (see, for example, [36], [37], [51], [317], [326]). However, a recent development of analysis on metric spaces, including fractals (see [22], [186], [187], [224]), has lead to emergence of other methods that are not linked so much to the local Euclidean structure of the underlying space.

Although singular spaces are not treated here, we still employ whenever possible those methods that could be applied also on such spaces. This desire has resulted in the abandonment of the parametrix method as well as the tools using smooth hypersurfaces such as the coarea formula and the boundary regularity of solutions, sometimes at expense of more technical arguments. Consequently, many proofs in this book are entirely new, even for the old well-known properties of the heat kernel and the Green function. A number of key theorems are presented with more than one proof, which should provide enough flexibility for building lecture courses for audiences with diverse background.

The material of Chapters 1 - 10, the first part of Chapter 11, and Chapter 13, belongs to the foundation of the subject. The rest of the book – the second part of Chapter 11, Chapters 12 and 14 - 16, contains more advanced results, obtained in the 1980s -1990s.

Let us briefly describe the contents of the individual chapters.

Chapters 1, 2, 6 contain the necessary material on the analysis in \mathbb{R}^n and the regularity theory of elliptic and parabolic equations in \mathbb{R}^n . They do not depend on the other chapters and can be either read independently or used as a reference source on the subject.

Chapter 3 contains a rather elementary introduction to Riemannian geometry, which focuses on the Laplace-Beltrami operator and the Green formula.

Chapter 4 introduces the Dirichlet Laplace operator as a self-adjoint operator in L^2 , which allows then to define the associated heat semigroup and to prove its basic properties. The spectral theorem is the main tool in this part.

Chapter 5 treats the Markovian properties of the heat semigroup, which amounts to the chain rule for the weak gradient, and the weak maximum principle for elliptic and parabolic problems. The account here does not use the smoothness of solutions; hence, the main tools are the Sobolev spaces.

Chapter 7 introduces the heat kernel on an arbitrary manifold as the integral kernel of the heat semigroup. The main tool is the regularity theory of Chapter 6, transplanted to manifolds. The existence of the heat kernel is derived from a local $L^2 \to L^{\infty}$ estimate of the heat semigroup, which in turn is a consequence of the Sobolev embedding theorem and the regularity theory. The latter implies also the smoothness of the heat kernel.

Chapter 8 deals with a number of issues related to the positivity or boundedness of solutions to the heat equation, which can be regarded as an extension of Chapter 5 using the smoothness of the solutions. It contains the results on the minimality of the heat semigroup and resolvent, the strong minimum principle for positive supersolutions, and some basic criteria for the stochastic completeness.

Chapter 9 treats the heat kernel as a fundamental solution. Based on that, some useful tools are introduced for verifying that a given function is the heat kernel, and some examples of heat kernels are given.

Chapter 10 deals with basic spectral properties of the Dirichlet Laplacian. It contains the variational principle for the bottom of the spectrum λ_1 , the positivity of the bottom eigenfunction, the discreteness of the spectrum and the positivity of λ_1 in relatively compact domains, and the characterization of the long time behavior of the heat kernel in terms of λ_1 .

Chapter 11 contains the material related to the use of the geodesic distance. It starts with the properties of Lipschitz functions, in particular, their weak differentiability, which allows then to use Lipschitz functions as test functions in various proofs. The following results are proved using the distance function: the essential self-adjointness of the Dirichlet Laplacian on geodesically complete manifolds, the volume tests for the stochastic completeness and parabolicity, and the estimates of the bottom of the spectrum.

Chapter 12 is the first of the four chapters dealing with upper bounds of the heat kernel. It contains the results on the integrated Gaussian estimates that are valid on an arbitrary manifold: the integrated maximum principle, the Davies-Gaffney inequality, Takeda's inequality, and some consequences. The proofs use the carefully chosen test functions based on the geodesic distance.

Chapter 13 is devoted to the Green function of the Laplace operator, which is constructed by integrating the heat kernel in time. Using the Green function together with the strong minimum principle allows to prove the local Harnack inequality for α -harmonic functions and its consequences – convergence theorems. As an example of application, the existence of the ground state on an arbitrary manifold is proved. Logically this Chapter belongs to the foundations of the subject and should have been placed much earlier in the sequence of the chapters. However, the proof of the local Harnack inequality requires one of the results of Chapter 12, which has necessitated the present order.

Chapter 14 deals with the on-diagonal upper bounds of the heat kernel, which requires additional hypothesis on the manifold in question. Normally such hypotheses are stated in terms of some isoperimetric or functional inequalities. We use here the approach based on the Faber-Krahn inequality for the bottom eigenvalue, which creates useful links with the spectral properties. The main result is that, to a certain extent, the on-diagonal upper bounds of the heat kernel are equivalent to the Faber-Krahn inequalities.

Chapter 15 continues the topic of the Gaussian estimates. The main technical result is Moser's mean-value inequality for solutions of the heat equation, which together with the integrated maximum principle allows to obtain pointwise Gaussian upper bounds of the heat kernel. We consider such estimates in the following three settings: arbitrary manifolds, the manifolds with the global Faber-Krahn inequality, and the manifolds with the relative Faber-Krahn inequality that leads to the Li-Yau estimates of the heat kernel.

Chapter 16 introduces alternative tools to deal with the Gaussian estimates. The main point is that the Gaussian upper bounds can be deduced directly from the on-diagonal upper bounds, although in a quite elaborate manner. As an application of these techniques, some on-diagonal lower estimates are proved.

Finally, Appendix A contains some reference material as was already mentioned above.

ACKNOWLEDGMENTS. The book was typeset in $\mathbb{IAT}_{\mathbb{E}}X$ using an excellent editor *Scientific Workplace* by TCI Software Research and MacKichan Software.

In the process of writing this book I was affiliated (permanently or temporarily) with the following institutions: Imperial College London, Institute of Henry Poincaré Paris, Chinese University of Hong Kong, Research Institute of Mathematical Sciences Kyoto, Institute of Control Sciences Moscow, University of Bielefeld, and ETH Zurich, with the support of the appropriate funding bodies.

However, the major part of the book was written during my three stays, totalling to twelve months, at the Institute of Mathematical Sciences of the Chinese University of Hong Kong, and I am very grateful to Professor Shing-Tung Yau for giving me that excellent opportunity. His support and encouragement have been paramount to me at all stages of my work.

Writing about maximum principle brings up memories of my teacher Eugene Landis. His masterful use of maximum principles has never been surpassed. My entire education in Analysis was hugely influenced by Landis, which has left an imprint on the style and choice of the material for this book.

A special thank is due to the late Serge Lang for useful discussions of the structure of the book.

It is a great pleasure to thank those colleagues of mine who have fruitfully affected my work in various ways: Martin Barlow, Alexander Bendikov, Isaac Chavel, Thierry Coulhon, Józef Dodziuk, Brian Davies, Wolfhard Hansen, Elton Pei Hsu, Jiaxin Hu, Vladimir Kondratiev, Takashi Kumagai, Ka-Sing Lau, Peter Li, Terry Lyons, Vladimir Maz'ya, Minoru Murata, Nikolai Nadirashvili, Michael Röckner, Laurent Saloff-Coste, Theo Sturm, Nina Ural'tseva.

Last but not least I am indebted to my family and especially to my wife Tatiana for inspiration and support.

Alexander Grigor'yan London - Paris - Hong Kong - Kyoto - Moscow - Bielefeld - Zurich 2002-2009

Bibliography

- 1. Adams R., "Sobolev spaces", Academic Press, 1975.
- Alexopoulos G.K., A lower estimate for central probabilities on polycyclic groups, Can. J. Math., 44 (1992) no.5, 897-910.
- 3. Alexopoulos G.K., An application of Homogenization theory to Harmonic analysis on solvable Lie groups of polynomial growth, *Pacific J. Math.*, **159** (1993) 19-45.
- Ancona A., Negatively curved manifolds, elliptic operators and the Martin boundary, Ann. of Math., 125 (1987) 495-536.
- Ancona A., Théorie du potentiel sur des graphes et des variétés, in: "Cours de l'Ecole d'été de probabilités de Saint-Flour, 1988", Lecture Notes Math. 1427, Springer, 1990. 4-112.
- Anderson M.T., Schoen R., Positive harmonic function on the complete manifolds of negative curvature, Ann. of Math., 121 (1983) 429-461.
- Anker J-Ph., Ji L., Heat kernel and Green function estimates on non-compact symmetric spaces, Geom. Funct. Anal., 9 (1999) 1035-1091.
- 8. Anker J-Ph., Ostellari P., The heat kernel on noncompact symmetric spaces, *in:* "Lie groups and symmetric spaces", Amer. Math. Soc. Transl. Ser. 2, 210, (2003) 27–46.
- Aronson D.G., Bounds for the fundamental solution of a parabolic equation, Bull. Amer. Math. Soc., 73 (1967) 890-896.
- Aronson D.G., Non-negative solutions of linear parabolic equations, Ann. Scuola Norm. Sup. Pisa. Cl. Sci. (4), 22 (1968) 607-694. Addendum 25 (1971) 221-228.
- Ashbaugh M. S., Isoperimetric and universal inequalities for eigenvalues, in: "Spectral theory and geometry (Edinburgh, 1998)", 95–139. London Math. Soc. Lecture Note Ser. 273, Cambridge Univ. Press, Cambridge, 1999.
- Atiyah M., Bott R., Patodi V. K., On the heat equation and the index theorem, Invent. Math., 19 (1973) 279–330. Errata Invent. Math., 28 (1975) 277–280.
- Auscher P., Coulhon T., Gaussian lower bounds for random walks from elliptic regularity, Ann. Inst. Henri Poincaré - Probabilités et Statistiques, 35 (1999) no.5, 605-630.
- 14. Auscher P., Coulhon T., Grigor'yan A., ed., "Heat kernels and analysis on manifolds, graphs, and metric spaces", Contemporary Mathematics 338, AMS, 2003.
- 15. Auscher P., Hofmann S., Lacey M., McIntosh A., Tchamitchian Ph., The solution of the Kato square root problem for second order elliptic operators on \mathbb{R}^n , Ann. of Math., **156** (2002) no.2, 633–654.
- Azencott R., Behavior of diffusion semi-groups at infinity, Bull. Soc. Math. (France), 102 (1974) 193-240.
- Bakry D., Coulhon T., Ledoux M., Saloff-Coste L., Sobolev inequalities in disguise, Indiana Univ. Math. J., 124 (1995) no.4, 1033-1074.
- Bakry D., Qian Z. M., Harnack inequalities on a manifold with positive or negative Ricci curvature, *Revista Matemática Iberoamericana*, 15 (1999) no.1, 143–179.
- 19. Barbatis G., Davies E.B., Sharp bounds on heat kernels of higher order uniformly elliptic operators, J. Operator Theory, **36** (1996) 179–198.

- Barbatis G., Filippas S., Tertikas A., Series expansion for L^p Hardy inequalities, Indiana Univ. Math. J., 52 (2003) 171-190.
- Barbatis G., Filippas S., Tertikas A., Critical heat kernel estimates for Schrödinger operators via Hardy-Sobolev inequalities, J. Funct. Anal., 208 (2004) 1-30.
- Barlow M.T., Diffusions on fractals, *in:* "Lectures on Probability Theory and Statistics, Ecole d'été de Probabilités de Saint-Flour XXV - 1995", Lecture Notes Math. 1690, Springer, 1998. 1-121.
- Barlow M.T., Heat kernels and sets with fractal structure, in: "Heat kernels and analysis on manifolds, graphs, and metric spaces", Contemporary Mathematics, 338 (2003) 11-40.
- 24. Barlow M.T., Bass R.F., Stability of parabolic Harnack inequalities, to appear in Trans. Amer. Math. Soc.
- Barlow M.T., Coulhon T., Grigor'yan A., Manifolds and graphs with slow heat kernel decay, Invent. Math., 144 (2001) 609-649.
- Barlow M.T., Coulhon T., Kumagai T., Characterization of sub-Gaussian heat kernel estimates on strongly recurrent graphs, Comm. Pure Appl. Math., 58 (2005) no.12, 1642–1677.
- 27. Barlow M.T., Grigor'yan A., Kumagai T., Heat kernel upper bounds for jump processes and the first exit time, J. Reine Angew. Math., 626 (2008) 135-157.
- Batty C.J.K., Asymptotic stability of Schrödinger semigroups: path integral methods, Math. Ann., 292 (1992) no.3, 457-492.
- Beltrami E., Delle variabili complesse sopra una superficie qualunque, Ann. Mat. Pura Appl. (2), 1 (1867) 329-366.
- Ben Arous G., Gradinaru M., Singularities of hypoelliptic Green functions, Potential Analysis, 8 (1998) 217-258.
- Ben Arous G., Léandre R., Décroissance exponentielle du noyau de la chaleur sur la diagonale 1,2, Prob. Th. Rel. Fields, 90 (1991) 175-202 and 377-402.
- Bendikov A., "Potential theory on infinite-dimensional Ablelian groups", de Gruyter Studies in Mathematics 21, De Gruyter, 1995.
- Bendikov A., Saloff-Coste L., Elliptic diffusions on infinite products, J. Reine Angew. Math., 493 (1997) 171–220.
- Bendikov A., Saloff-Coste L., On- and off-diagonal heat kernel behaviors on certain infinite dimensional local Dirichlet spaces, Amer. J. Math., 122 (2000) 1205– 1263.
- Benjamini I., Chavel I., Feldman E.A., Heat kernel lower bounds on Riemannian manifolds using the old ideas of Nash, Proceedings of London Math. Soc., 72 (1996) 215-240.
- Berger M., Gauduchon P., Mazet E., Le spectre d'une variété riemanniennes, Lecture Notes Math. 194, Springer, 1971.
- 37. Berline N., Getzler E., Vergne M., Heat kernels and Dirac operators, Grundlehren der mathematischen Wissenschaften 298, Springer, 1992.
- Bers L., John F., Schechter M., "Partial differential equations", John Wiley & Sons (Interscience), 1964.
- 39. Beurling A., Deny J., Dirichlet spaces, Proc. Nat. Acad. Sci. USA, 45 (1959) 208-215.
- Biroli M., Mosco U., Sobolev and isoperimetric inequalities for Dirichlet forms on homogeneous spaces, Rend. Mat. Acc. Lincei, s.9, 6 (1995) 37-44.
- 41. Blumental R.M., Getoor R.K., "Markov processes and potential theory", Academic Press, New York, 1968.
- 42. Bobkov S.G., Houdré C., Some connections between isoperimetric and Sobolevtype inequalities, Mem. Amer. Math. Soc., **129** (1997) no. 616, viii+111pp.

- Bobkov S.G., Ledoux M., Poincaré's nequalities and Talagrand's concentration phenomenon for the exponential distribution, *Probab. Theory Relat. Fields*, 107 (1997) no.3, 383-400.
- 44. Bombieri E., de Giorgi E., Miranda M., Una maggiorazioni a-priori relative alle ipersuperfici minimali non parametriche, Arch. Rat. Mech. Anal., **32** (1969) 255-367.
- 45. Boothby W. M., "An introduction to differentiable manifolds and Riemannian geometry", Pure and Applied Mathematics 120, Academic Press, 1986.
- Boukricha A., Das Picard-Prinzip und verwandte Fragen bei Störung von harmonischen Räumen, Math. Ann., 239 (1979) 247-270.
- 47. Brooks R., A relation between growth and the spectrum of the Laplacian, Math. Z., 178 (1981) 501-508.
- 48. Buser P., A note on the isoperimetric constant, Ann. Sci. Ecole Norm. Sup., 15 (1982) 213-230.
- Carlen E.A., Kusuoka S., Stroock D.W., Upper bounds for symmetric Markov transition functions, Ann. Inst. H. Poincaré Probab. Statist., 23 (1987) no.2, suppl. 245-287.
- 50. Carron G., Inégalités isopérimétriques de Faber-Krahn et conséquences, in: "Actes de la table ronde de géométrie différentielle (Luminy, 1992)", Collection SMF Séminaires et Congrès 1, 1996. 205–232.
- 51. Chavel I., "Eigenvalues in Riemannian geometry", Academic Press, New York, 1984.
- Chavel I., "Riemannian geometry : a modern introduction", Cambridge Tracts in Mathematics 108, Cambridge University Press, 1993.
- 53. Chavel I., "Isoperimetric inequalities: differential geometric and analytic perspectives", Cambridge Tracts in Mathematics 145, Cambridge University Press, 2001.
- Chavel I., Feldman E.A., Isoperimetric constants, the geometry of ends, and large time heat diffusion in Riemannian manifolds, Proc London Math. Soc., 62 (1991) 427-448.
- 55. Chavel I., Feldman E.A., Modified isoperimetric constants, and large time heat diffusion in Riemannian manifolds, *Duke Math. J.*, **64** (1991) no.3, 473-499.
- Cheeger J., A lower bound for the smallest eigenvalue of the Laplacian, in: "Problems in Analysis: A Symposium in honor of Salomon Bochner", Princeton University Press. Princeton, 1970, 195-199.
- 57. Cheeger J., Gromov M., Taylor M., Finite propagation speed, kernel estimates for functions of the Laplace operator, and the geometry of complete Riemannian manifolds, J. Diff. Geom., 17 (1982) 15-53.
- Cheeger J., Yau S.-T., A lower bound for the heat kernel, Comm. Pure Appl. Math., 34 (1981) 465-480.
- Cheng S.Y, Li P., Heat kernel estimates and lower bounds of eigenvalues, Comment. Mat. Helv., 56 (1981) 327-338.
- Cheng S.Y., Eigenvalue comparison theorem and its geometric application, Math. Z., 143 (1975) 289-297.
- Cheng S.Y., Li P., Yau S.-T., On the upper estimate of the heat kernel of a complete Riemannian manifold, Amer. J. Math., 103 (1981) no.5, 1021-1063.
- Cheng S.Y., Yau S.-T., Differential equations on Riemannian manifolds and their geometric applications, Comm. Pure Appl. Math., 28 (1975) 333-354.
- Chernoff P.R., Essential self-adjointness of powers of generators of hyperbolic equations, J. Funct. Anal., 12 (1973) 401–414.
- Chung F.R.K., "Spectral Graph Theory", CBMS Regional Conference Series in Mathematics 92, AMS publications, 1996.
- Chung F.R.K., Grigor'yan A., Yau S.-T., Upper bounds for eigenvalues of the discrete and continuous Laplace operators, Advances in Math., 117 (1996) 165-178.

- Chung F.R.K., Grigor'yan A., Yau S.-T., Eigenvalues and diameters for manifolds and graphs, *in:* "Tsing Hua Lectures on Geometry and Analysis", Ed. S.-T.Yau, International Press, 1997. 79-105.
- Chung F.R.K., Grigor'yan A., Yau S.-T., Higher eigenvalues and isoperimetric inequalities on Riemannian manifolds and graphs, *Comm. Anal. Geom*, 8 (2000) no.5, 969-1026.
- Chung F.R.K., Yau S.-T., Eigenvalue inequalities for graphs and convex subgraphs, Comm. in Analysis and Geom., 2 (1994) 628-639.
- Chung F.R.K., Yau S.-T., A Harnack inequality for Dirichlet eigenvalues, J. Graph Theory, 34 (2000) no.4, 247–257.
- Chung F.R.K., Yau, S.-T., Eigenvalues of graphs and Sobolev inequalities, Combinatorics, Probability and Computing, 4 (1995) 11-26.
- Chung K.L., Zhao Z., "From Brownian motion to Schrödinger's equation", A series of Comprehensive studies in Mathematics 312, Springer, 1985.
- Chung L.O., Existence of harmonic L¹ functions in complete Riemannian manifolds, Proc. Amer. Math. Soc., 88 (1983) 531-532.
- 73. Conway J., "A course in functional analysis", Graduate Texts in Mathematics 96, Springer-Verlag, New York, 1990.
- Coulhon T., Noyau de la chaleur et discrétisation d'une variété riemannienne, Israël J. Math., 80 (1992) 289-300.
- Coulhon T., Iteration de Moser et estimation Gaussienne du noyau de la chaleur, J. Operator Theory, 29 (1993) 157-165.
- Coulhon T., Dimensions at infinity for Riemannian manifolds, Potential Anal., 4 (1995) no.5, 335-344.
- 77. Coulhon T., Ultracontractivity and Nash type inequalities, J. Funct. Anal., 141 (1996) no.2, 510-539.
- Coulhon T., Off-diagonal heat kernel lower bounds without Poincaré, J. London Math. Soc., 68 (2003) no.3, 795–816.
- Coulhon T., Grigor'yan A., Heat kernel, volume growth and anti-isoperimetric inequalities, C.R. Acad. Sci. Paris, Sér. I Math., 322 (1996) 1027-1032.
- Coulhon T., Grigor'yan A., On-diagonal lower bounds for heat kernels on noncompact manifolds and Markov chains, *Duke Math. J.*, 89 (1997) no.1, 133-199.
- Coulhon T., Grigor'yan A., Random walks on graphs with regular volume growth, Geom. Funct. Anal., 8 (1998) 656-701.
- Coulhon T., Grigor'yan A., Pointwise estimates for transition probabilities of random walks on infinite graphs, *in:* "Fractals in Graz 2001", Ed. P.Grabner, W.Woess, Trends in Mathematics, Birkäuser, 2003. 119-134.
- Coulhon T., Grigor'yan A., Levin D., On isoperimetric profiles of product spaces, Comm. Anal. Geom., 11 (2003) no.1, 85-120.
- Coulhon T., Grigor'yan A., Pittet Ch., A geometric approach to on-diagonal heat kernel lower bounds on groups, Ann. Inst. Fourier, Grenoble, 51 (2001) no.6, 1763-1827.
- Coulhon T., Grigor'yan A., Zucca, F, The discrete integral maximum principle and its applications, *Tohoku Math. J.*, 57 (2005) no.4, 559-587.
- Coulhon T., Saloff-Coste L., Isopérimétrie pour les groupes et les variétés, Revista Matemática Iberoamericana, 9 (1993) no.2, 293-314.
- Coulhon T., Sikora A., Gaussian heat kernel upper bounds via Phragmén-Lindelöf theorem, Proc. Lond. Math. Soc., 96 (2008) no.2, 507–544.
- Courant R., Hilbert D., "Methods of Mathematical Physics, Vol. 1", Interscience Publishers, 1953.
- 89. Courant R., Hilbert D., "Methods of Mathematical Physics, Vol. 1I: Partial differential equations", Interscience Publishers, 1962.

- Cranston M., A probabilistic approach to Martin boundaries for manifolds with end, Prob. Theory and Related Fields, 96 (1993) 319–334.
- Cranston M., Greven A., Coupling and harmonic functions in the case of continuous time Markov processes, Stochastic Processes and their Applications, 60 (1995) no.2, 261-286.
- 92. Davies E.B., "One-parameter semigroups", Academic Press, 1980.
- Davies E.B., L¹ properties of second order elliptic operators, Bull. London Math. Soc., 17 (1985) no.5, 417-436.
- Davies E.B., Explicit constants for Gaussian upper bounds on heat kernels, Amer. J. Math., 109 (1987) 319-334.
- 95. Davies E.B., Gaussian upper bounds for the heat kernel of some second-order operators on Riemannian manifolds, J. Funct. Anal., 80 (1988) 16-32.
- 96. Davies E.B., "Heat kernels and spectral theory", Cambridge University Press, 1989.
- Davies E.B., Heat kernel bounds, conservation of probability and the Feller property, J. d'Analyse Math., 58 (1992) 99-119.
- Davies E.B., The state of art for heat kernel bounds on negatively curved manifolds, Bull. London Math. Soc., 25 (1993) 289-292.
- Davies E.B., Large deviations for heat kernels on graphs, J. London Math. Soc. (2), 47 (1993) 65-72.
- Davies E.B., L^p spectral theory of higher-order elliptic differential operators, Bull. London Math. Soc., 29 (1997) 513-546.
- 101. Davies E.B., Non-Gaussian aspects of heat kernel behaviour, J. London Math. Soc., 55 (1997) no.1, 105-125.
- 102. Davies E.B., Mandouvalos N., Heat kernel bounds on hyperbolic space and Kleinian groups, *Proc. London Math. Soc.*(3), **52** (1988) no.1, 182-208.
- 103. De Giorgi E., Sulla differenziabilita e l'analiticita delle estremali degli integrali multipli regolari, Mem. Accad. Sci. Torino Cl. Sci. Fis. Mat. Nat., ser.3, 3 (1957) 25-43.
- 104. Debiard A., Gaveau B., Mazet E., Théorèmes de comparison in gèomètrie riemannienne, Publ. Kyoto Univ., 12 (1976) 391-425.
- 105. Dekkers S.A.J., Finite propagation speed for solutions of the parabolic *p*-Laplace equation on manifolds, *Comm. Anal. Geom.*, **14** (2005) no.4, 741-768.
- 106. Deny J., Les potentiels d'energie finite, Acta Math., 82 (1950) 107-183.
- 107. DiBenedetto E., "Degenerate parabolic equations", Universitext, Springer, 1993.
- 108. Dodziuk J., Maximum principle for parabolic inequalities and the heat flow on open manifolds, *Indiana Univ. Math. J.*, **32** (1983) no.5, 703-716.
- 109. Donnelly H., Bounded harmonic functions and positive Ricci curvature, Math. Z., 191 (1986) 559-565.
- 110. Doyle P.G., On deciding whether a surface is parabolic or hyperbolic, *Contemporary Mathematics*, **73** (1988) 41-49.
- Doyle P.G., Snell J.L., "Random walks and electric networks", Carus Mathematical Monographs 22, Mathematical Association of America, Washington, DC, 1984.
- 112. Driver B.K., Heat kernels measures and infinite dimensional analysis, in: "Heat kernels and analysis on manifolds, graphs, and metric spaces", Contemporary Mathematics, 338 (2003) 101-142.
- Driver B.K., Gordina M., Heat kernel analysis on infinite-dimensional Heisenberg groups, J. Funct. Anal., 255 (2008) no.9, 2395–2461.
- Driver B.K., Melcher T., Hypoelliptic heat kernel inequalities on the Heisenberg group, J. Funct. Anal., 221 (2005) no.2, 340–365.
- 115. Dynkin E.B., "Markov processes", Springer, 1965.

- 116. Ecker K., Huisken G., Mean curvature evolution of entire graphs, Ann. Math.,
 130 (1989) no.3, 453–471.
- Escauriaza L., Bounds for the fundamental solution of elliptic and parabolic equations in nondivergence form, Comm. Partial Differential Equations, 25 (2000) 821-845.
- Evans L.C., "Partial differential equations", Graduate Studies in Mathematics 19, AMS, 1997.
- Evans S.N., Multiple points in the sample paths of a Lévy process, Probab. Th. Rel. Fields, 76 (1987) 359-367.
- 120. Fabes E.B., Stroock D.W., A new proof of Moser's parabolic Harnack inequality via the old ideas of Nash, Arch. Rat. Mech. Anal., 96 (1986) 327-338.
- 121. Friedman A., "Partial differential equations", Holt, Rinehart, and Winston, 1969.
- 122. Friedrichs K.O., The identity of weak and strong extensions of differential operators, Trans. AMS, 55 (1944) 132-151.
- Friedrichs K.O., On the differentiability of the solutions of linear elliptic differential equations, Comm. Pure Appl. Math. 6 (1953) 299-326.
- 124. Fukushima M., Oshima Y., Takeda M., "Dirichlet forms and symmetric Markov processes", Studies in Mathematics 19, De Gruyter, 1994.
- 125. Fukushima M., Uemura T., Capacitary bounds of measures and ultracontractivity of time changed processes, J. Math. Pures Appl. (9), 82 (2003) no.5, 553–572.
- 126. Gaffney M.P., A special Stokes theorem for complete Riemannian manifolds, Annals of Math., 60 (1954) 140-145.
- 127. Gaffney M.P., The conservation property of the heat equation on Riemannian manifolds, Comm. Pure Appl. Math., 12 (1959) 1-11.
- Garofalo N., Nhieu D.-M., Isoperimetric and Sobolev inequalities for Carnot-Caratheodory spaces and the existence of minimal surfaces, *Comm. Pure Appl. Math.*, 49 (1996) 1081-1144.
- 129. Gerl P., Random walks on graphs, Lecture Notes Math. 1210, Springer, 1986. 285-303.
- Gilbarg D., Trudinger N., "Elliptic partial differential equations of second order", Springer, 1977,1983,1998,2001.
- 131. Gilkey P.B., "The index theorem and the heat equation", Publish or Perish, Boston MA, 1974.
- 132. Gilkey P.B., "Invariance theory, heat equation and the Atiyah-Singer index theory", CRC Press, Studies in Advanced Mathematics, 1995.
- 133. Greene R., Wu W., "Function theory of manifolds which possess a pole", Lecture Notes Math. 699, Springer, 1979.
- 134. Grigor'yan A., On the existence of a Green function on a manifold, (in Russian) Uspekhi Matem. Nauk, 38 (1983) no.1, 161-162. Engl. transl.: Russian Math. Surveys, 38 (1983) no.1, 190-191.
- 135. Grigor'yan A., Isoperimetric inequalities for Riemannian products, (in Russian) Mat. Zametki, 38 (1985) no.4, 617-626. Engl. transl.: Math. Notes, 38 (1985) 849-854.
- 136. Grigor'yan A., On the existence of positive fundamental solution of the Laplace equation on Riemannian manifolds, (in Russian) Matem. Sbornik, 128 (1985) no.3, 354-363. Engl. transl.: Math. USSR Sb., 56 (1987) 349-358.
- 137. Grigor'yan A., On stochastically complete manifolds, (in Russian) DAN SSSR,
 290 (1986) no.3, 534-537. Engl. transl.: Soviet Math. Dokl., 34 (1987) no.2, 310-313.
- 138. Grigor'yan A., A criterion for the existence on a Riemannian manifold of a nontrivial bounded harmonic function with finite Dirichlet integral, (in Russian) Dokl. Akad. Nauk SSSR, 293 (1987) no.3, 529-531. Engl. transl.: Soviet Math. Dokl, 35 (1987) no.2, 339-341.

- Grigor'yan A., On Liouville theorems for harmonic functions with finite Dirichlet integral, (in Russian) Matem. Sbornik, 132 (1987) no.4, 496-516. Engl. transl.: Math. USSR Sbornik, 60 (1988) no.2, 485-504.
- 140. Grigor'yan A., Set of positive solutions of Laplace-Beltrami equation on special type of Riemannian manifolds, (in Russian) Izv. Vyssh. Uchebn. Zaved., Matematika, (1987) no.2, 30-37. Engl. transl.: Soviet Math (Iz.VUZ), 31 (1987) no.2, 48-60.
- 141. Grigor'yan A., On the fundamental solution of the heat equation on an arbitrary Riemannian manifold, (in Russian) Mat. Zametki, 41 (1987) no.3, 687-692. Engl. transl.: Math. Notes, 41 (1987) no.5-6, 386-389.
- 142. Grigor'yan A., Stochastically complete manifolds and summable harmonic functions, (in Russian) *Izv. AN SSSR*, ser. matem., **52** (1988) no.5, 1102-1108. Engl. transl.: *Math. USSR Izvestiya*, **33** (1989) no.2, 425-432.
- 143. Grigor'yan A., Bounded solutions of the Schrödinger equation on non-compact Riemannian manifolds, (in Russian) *Trudy seminara I.G.Petrovskogo*, (1989) no.14, 66-77. Engl. transl.: J. Soviet Math., **51** (1990) no.3, 2340-2349.
- 144. Grigor'yan A., Dimension of spaces of harmonic functions, (in Russian) Mat. Zametki, 48 (1990) no.5, 55-61. Engl. transl.: Math. Notes, 48 (1990) no.5, 1114-1118.
- 145. Grigor'yan A., The heat equation on non-compact Riemannian manifolds, (in Russian) Matem. Sbornik, 182 (1991) no.1, 55-87. Engl. transl.: Math. USSR Sb., 72 (1992) no.1, 47-77.
- 146. Grigor'yan A., Heat kernel upper bounds on a complete non-compact manifold, Revista Matemática Iberoamericana, 10 (1994) no.2, 395-452.
- 147. Grigor'yan A., Integral maximum principle and its applications, Proc. Roy. Soc. Edinburgh, 124A (1994) 353-362.
- Grigor'yan A., Heat kernel on a manifold with a local Harnack inequality, Comm. Anal. Geom., 2 (1994) no.1, 111-138.
- 149. Grigor'yan A., Heat kernel on a non-compact Riemannian manifold, in: "1993 Summer Research Institute on Stochastic Analysis", Ed. M.Pinsky et al., Proceedings of Symposia in Pure Mathematics, 57 (1995) 239-263.
- 150. Grigor'yan A., Upper bounds of derivatives of the heat kernel on an arbitrary complete manifold, J. Funct. Anal., 127 (1995) no.2, 363-389.
- Grigor'yan A., Gaussian upper bounds for the heat kernel on arbitrary manifolds, J. Diff. Geom., 45 (1997) 33-52.
- Grigor'yan A., Escape rate of Brownian motion on weighted manifolds, Applicable Analysis, 71 (1999) no.1-4, 63-89.
- Grigor'yan A., Isoperimetric inequalities and capacities on Riemannian manifolds, Operator Theory: Advances and Applications, 109 (1999) 139-153.
- 154. Grigor'yan A., Estimates of heat kernels on Riemannian manifolds, in: "Spectral Theory and Geometry. ICMS Instructional Conference, Edinburgh 1998", Ed. E.B. Davies and Yu. Safarov, London Math. Soc. Lecture Note Series 273, Cambridge Univ. Press, 1999. 140-225.
- 155. Grigor'yan A., Analytic and geometric background of recurrence and nonexplosion of the Brownian motion on Riemannian manifolds, Bull. Amer. Math. Soc., 36 (1999) 135-249.
- 156. Grigor'yan A., Heat kernels and function theory on metric measure spaces, in: "Heat kernels and analysis on manifolds, graphs, and metric spaces", Contemporary Mathematics, 338 (2003) 143-172.
- 157. Grigor'yan A., Heat kernels on weighted manifolds and applications, *Contemporary Mathematics*, **398** (2006) 93-191.
- 158. Grigor'yan A., Heat kernels on metric measure spaces, to appear in Handbook of Geometric Analysis No.2 Ed. L. Ji, P. Li, R. Schoen, L. Simon, Advanced Lectures in Math., International Press, 2008.

- 159. Grigor'yan A., Hansen W., A Liouville property for Schrödinger operators, Math. Ann., 312 (1998) 659-716.
- Grigor'yan A., Hansen W., Lower estimates for a perturbed Green function, J. d'Analyse Math., 104 (2008) 25-58.
- 161. Grigor'yan A., Hsu, Elton P., Volume growth and escape rate of Brownian motion on a Cartan-Hadamard manifold, *in:* "Sobolev Spaces in Mathematics II", Ed. V. Maz'ya, International Mathematical Series 9, Springer, 2009. 209-225.
- 162. Grigor'yan A., Hu J., Off-diagonal upper estimates for the heat kernel of the Dirichlet forms on metric spaces, *Invent. Math.*, 174 (2008) 81-126.
- 163. Grigor'yan A., Hu J., Upper bounds of heat kernels on doubling spaces, preprint
- 164. Grigor'yan A., Hu J., Lau K.S., Heat kernels on metric-measure spaces and an application to semi-linear elliptic equations, Trans. Amer. Math. Soc., 355 (2003) no.5, 2065-2095.
- 165. Grigor'yan A., Hu J., Lau K.S., Equivalence conditions for on-diagonal upper bounds of heat kernels on self-similar spaces, J. Funct. Anal., 237 (2006) 427-445.
- 166. Grigor'yan A., Hu J., Lau K.S., Obtaining upper bounds of heat kernels from lower bounds, Comm. Pure Appl. Math., 61 639-660.
- 167. Grigor'yan A., Hu J., Lau K.S., Heat kernels on metric spaces with doubling measure, to appear in *Proceedings of Fractal Conference in Greifswald IV*
- 168. Grigor'yan A., Kelbert M., Range of fluctuation of Brownian motion on a complete Riemannian manifold, Ann. Prob., 26 (1998) 78-111.
- 169. Grigor'yan A., Kelbert M., On Hardy-Littlewood inequality for Brownian motion on Riemannian manifolds, J. London Math. Soc. (2), 62 (2000) 625-639.
- 170. Grigor'yan A., Kelbert M., Asymptotic separation for independent trajectories of Markov processes, Prob. Th. Rel. Fields, 119 (2001) no.1, 31-69.
- 171. Grigor'yan A., Kelbert M., Recurrence and transience of branching diffusion processes on Riemannian manifolds, *Annals of Prob.*, **31** (2003) no.1, 244-284.
- 172. Grigor'yan A., Kumagai T., On the dichotomy in the heat kernel two sided estimates, Proceedings of Symposia in Pure Mathematics, 77 Amer. Math. Soc., Providence, RI, (2008) 199-210.
- 173. Grigor'yan A., Nadirashvili N.S., Liouville theorems and exterior boundary value problems, (in Russian) *Izv. Vyssh. Uchebn. Zaved.*, *Matematika*, (1987) no.5, 25-33. Engl. transl.: Soviet Math. (*Iz.VUZ*), **31** (1987) no.5, 31-42.
- 174. Grigor'yan A., Netrusov Yu., Yau S.-T., Eigenvalues of elliptic operators and geometric applications, *in:* "Eigenvalues of Laplacians and other geometric operators", Surveys in Differential Geometry IX, (2004) 147-218.
- 175. Grigor'yan A., Noguchi M., The heat kernel on hyperbolic space, Bull. London Math. Soc., 30 (1998) 643-650.
- 176. Grigor'yan A., Saloff-Coste L., Heat kernel on connected sums of Riemannian manifolds, Math. Research Letters, 6 (1999) no.3-4, 307-321.
- 177. Grigor'yan A., Saloff-Coste L., Dirichlet heat kernel in the exterior of a compact set, Comm. Pure Appl. Math, 55 (2002) 93-133.
- 178. Grigor'yan A., Saloff-Coste L., Hitting probabilities for Brownian motion on Riemannian manifolds, J. Math. Pures et Appl., 81 (2002) 115-142.
- 179. Grigor'yan A., Saloff-Coste L., Stability results for Harnack inequalities, Ann. Inst. Fourier, Grenoble, 55 (2005) no.3, 825-890.
- Grigor'yan A., Saloff-Coste L., Heat kernel on manifolds with ends, Ann. Inst. Fourier, Grenoble, 59 (2009)
- 181. Grigor'yan A., Telcs A., Sub-Gaussian estimates of heat kernels on infinite graphs, Duke Math. J., 109 (2001) no.3, 452-510.
- 182. Grigor'yan A., Telcs A., Harnack inequalities and sub-Gaussian estimates for random walks, Math. Ann., **324** no.3, (2002) 521-556.

- 183. Grigor'yan A., Yau S.-T., Decomposition of a metric space by capacitors, in: "Differential Equations: La Pietra 1996", Ed. Giaquinta et. al., Proceeding of Symposia in Pure Mathematics, 65 1999. 39-75.
- 184. Grigor'yan A., Yau S.-T., Isoperimetric properties of higher eigenvalues of elliptic operator, Amer. J. Math, 125 (2003) 893-940.
- 185. Grigor'yan A., Yau S.-T., ed., "Eigenvalues of Laplacians and other geometric operators", Surveys in Differential Geometry IX, International Press, 2004.
- 186. **Gromov M.**, "Structures métriques pour les variétés riemanniennes", Paris: Cedic/Ferdnand Nathan, 1981.
- 187. Gromov M., "Metric structures for Riemannian and non-Riemannian spaces", Birkhäuser, 1999,2007.
- 188. Gross L., Logarithmic Sobolev inequalities, Amer. J. Math., 97 (1976) 1061-1083.
- 189. Gross L., Heat kernel analysis on Lie groups, in: "Stochastic analysis and related topics, VII (Kusadasi, 1998)", Progr. Probab. 48, Birkhäuser Boston, Boston, MA, 2001. 1–58.
- 190. Guionnet A., Zegarlinski B., Lectures on logarithmic Sobolev inequalities, in: "Séminaire de Probabilités, XXXVI", Lecture Notes in Math. 1801, Springer, Berlin, yab2003 1–134.
- 191. Guivarc'h Y., Sur la loi des grands nombres et le rayon spectral d'une marche aléatoire, Journée sur les marches aléatoires, Astérisque, 74 (1980) 47-98.
- 192. Gushchin A.K., On the uniform stabilization of solutions of the second mixed problem for a parabolic equation, (in Russian) Matem. Sbornik, 119(161) (1982) no.4, 451-508. Engl. transl.: Math. USSR Sb., 47 (1984) 439-498.
- 193. Gushchin A.K., Michailov V.P., Michailov Ju.A., On uniform stabilization of the solution of the second mixed problem for a second order parabolic equation, (in Russian) Matem. Sbornik, 128(170) (1985) no.2, 147-168. Engl. transl.: Math. USSR Sb., 56 (1987) 141-162.
- 194. Halmos P.R., "Measure theory", D. Van Nostrand Company, Inc., New York, N. Y., 1950.
- 195. Hambly B.M., Kumagai T., Transition density estimates for diffusion processes on post critically finite self-similar fractals, *Proc. London Math. Soc.* (3), 78 (1999) 431–458.
- 196. Hamel F., Nadirashvili N., Russ E., Some isoperimetric problems for the principal eigenvalues of second-order elliptic operators in divergence form, C. R. Acad. Sci. Paris, Ser. I, 344 (2007) 169–174.
- 197. Hansen W., Harnack inequalities for Schrödinger operators, Ann. Scuola Norm. Sup. Pisa, 28 (1999) 413-470.
- 198. Hebisch W., Saloff-Coste, L., Gaussian estimates for Markov chains and random walks on groups, Ann. Prob., 21 (1993) 673–709.
- 199. Hebisch W., Saloff-Coste, L., On the relation between elliptic and parabolic Harnack inequalities, Ann. Inst. Fourier, 51 (2001) no.5, 1437-1481.
- 200. Helgason S., "Differential geometry, Lie groups, and symmetric spaces", Graduate Studies in Mathematics 34, AMS, 2001.
- Herbst I.W., Zhao Z., Green's functions for the Schrödinger equation with shortrange potential, Duke Math. J., 59 (1989) 475–519.
- 202. Hino M., Ramírez, J., Small-time Gaussian behavior of symmetric diffusion semigroup, Ann. Prob., 31 (2003) no.3, 1254-1295.
- 203. Hoffman D., Spruck J., Sobolev and isoperimetric inequalities for Riemannian submanifolds, Comm. Pure Appl. Math., 27 (1974) 715–727. See also "A correction to: Sobolev and isoperimetric inequalities for Riemannian submanifolds", Comm. Pure Appl. Math., 28 (1975) no.6, 765–766.
- 204. Holopainen I., Rough isometries and *p*-harmonic functions with finite Dirichlet integral, *Revista Matemática Iberoamericana*, **10** (1994) 143-176.

- 205. Holopainen I., Solutions of elliptic equations on manifolds with roughly Euclidean ends, Math. Z., 217 (1994) 459-477.
- 206. Holopainen I., Volume growth, Green's functions and parabolicity of ends, Duke Math. J., 97 (1999) no.2, 319-346.
- 207. Hörmander L., "The Analysis of Linear Partial Differential Operators I", Springer, 1983.
- 208. Hörmander L., "The Analysis of Linear Partial Differential Operators III", Springer, 1985.
- 209. Hsu, Elton P., Heat semigroup on a complete Riemannian manifold, Ann. Probab.,
 17 (1989) 1248-1254.
- Hsu, Elton P., "Stochastic Analysis on Manifolds", Graduate Texts in Mathematics 38, 2002.
- 211. Ichihara K., Curvature, geodesics and the Brownian motion on a Riemannian manifold. I Recurrence properties, Nagoya Math. J., 87 (1982) 101-114.
- 212. Ichihara K., Curvature, geodesics and the Brownian motion on a Riemannian manifold. II Explosion properties, Nagoya Math. J., 87 (1982) 115-125.
- Isham C. J., "Modern differential geometry for physicists", World Scientific Lecture Notes in Physics 61, World Scientific Publishing, 1999.
- 214. Ishige K., Murata M., Uniqueness of nonnegative solutions of the Cauchy problem for parabolic equations on manifolds or domains, Ann. Scuola Norm. Sup. Pisa Cl. Sci. (4), 30 (2001) no.1, 171–223.
- 215. Jerison D., Sánchez-Calle A., Estimates for the heat kernel for a sum of squares of vector fields, Indiana Univ. Math. J., 35 (1986) no.4, 835-854.
- 216. Jerison D., Sánchez-Calle A., Subelliptic, second order differential operators, in: "Complex analysis, III (College Park, Md., 1985-86)", Lecture Notes Math. 1277, Springer, 1987. 46-77.
- 217. Jorgenson J., Lang S., The ubiquitous heat kernel, *in:* "Mathematics unlimited—2001 and beyond", Springer, Berlin, 2001. 655–683.
- 218. Jorgenson J., Lang S., Spherical inversion on SL₂(C), in: "Heat kernels and analysis on manifolds, graphs, and metric spaces", Contemporary Mathematics, 338 (2003) 241-270.
- 219. Jorgenson J., Lynne, W., editors, "The ubiquitous heat kernel. Papers from a special session of the AMS Meeting held in Boulder, CO, October 2–4, 2003", Contemporary Mathematics 398, American Mathematical Society, Providence, RI, 2006.
- 220. Kaimanovich V.A., Vershik A.M., Random walks on discrete groups: boundary and entropy, Ann. Prob., 11 (1983) no.3, 457-490.
- 221. Karp L., Subharmonic functions, harmonic mappings and isometric immersions, in: "Seminar on Differential Geometry", Ed. S.T.Yau, Ann. Math. Stud. 102, Princeton, 1982.
- 222. Karp L., Li P., The heat equation on complete Riemannian manifolds, unpublished manuscript 1983.
- 223. Khas'minskii R.Z., Ergodic properties of recurrent diffusion prossesses and stabilization of solution to the Cauchy problem for parabolic equations, *Theor. Prob. Appl.*, 5 (1960) no.2, 179-195.
- 224. Kigami J., "Analysis on fractals", Cambridge University Press, 2001.
- 225. Kigami J., Local Nash inequality and inhomogeneity of heat kernels, Proc. London Math. Soc. (3), 89 (2004) 525-544.
- 226. Kirillov A.A., Gvishiani A.D., "Theorems and problems in functional analysis", Springer-Verlag, New York-Berlin, 1982.
- 227. Klingenberg W., "Riemannian Geometry", de Gruyter, 1982.
- 228. Kobayashi S., Nomizu K., "Foundations of differential geometry. Vol.I, II", Interscience Publishers, New York, 1963,1969.

- 229. Kolmogorov A.N., Fomin S.V., "Elements of the theory of functions and functional analysis", (in Russian) Nauka, Moscow, 1989.
- 230. Kondrat'ev V.A., Landis E.M., Qualitative theory of linear second-order partial differential equations, (in Russian) Itogi Nauki i Techniki, serija Sovremennye Problemy Matematiki, Fundamental'nye Napravlenija 32, VINITI, Moscow, 1988. 99-215. Engl. transl.: in: "Partial Differential Equations III", Encyclopedia of Math. Sci. 32, Springer, 1990.
- 231. Koranyi A., Taylor J.C., Minimal solutions of the heat equation and uniqueness of the positive Cauchy problem on homogeneous spaces, *Proc. Amer. Math. Soc.*, 94 (1985) 273-278.
- 232. Korevaar N., Upper bounds for eigenvalues of conformal metric, J. Diff. Geom., 37 (1993) 73-93.
- 233. Kotani M., Shirai T., Sunada T., Asymptotic behavior of the transition probability of a random walk on an infinite graph, J. Funct. Anal., 159 (1998) 664-689.
- 234. Kotani M., Sunada T., Albanese maps and off diagonal long time asymptotics for the heat kernel, preprint 1999.
- 235. Kreyszig E., "Introductory functional analysis with applications", John Wiley & Sons, Inc., New York, 1989.
- 236. Krylov N.V., Safonov M.V., A certain property of solutions of parabolic equations with measurable coefficients, (in Russian) Izv. Akad. Nauk SSSR, 44 (1980) 81-98. Engl. transl.: Math. USSR Izvestija, 16 (1981) 151-164.
- 237. Kurdyukov Yu.A., L^p-theory of elliptic differential operators on manifolds of bounded geometry, Acta Applic. Math., 23 (1991) 223-260.
- 238. Kusuoka S., A dissusion process on a fractal, in: "Probabilistic methods in Mathematical Physics, Taniguchi Symp., Katana, 1985", Ed. K.Ito and N.Ikeda, Kinokuniya-North Holland, Amsterdam, 1987. 251-274.
- 239. Kusuoka S., Stroock D.W., Long time estimates for the heat kernel associated with uniformly subelliptic symmetric second order operators, Ann. Math., 127 (1989) 165-189.
- 240. Kuz'menko Yu.T., Molchanov S.A., Counterexamples to Liouville-type theorems, (in Russian) Vestnik Moskov. Univ. Ser. I Mat. Mekh., (1979) no.6, 39-43. Engl. transl.: Moscow Univ. Math. Bull., 34 (1979) 35-39.
- Ladyzenskaja O.A., V.A. Solonnikov, Ural'ceva N.N., "Linear and quasilinear equations of parabolic type", Providence, Rhode Island, 1968.
- 242. Ladyzhenskaya O.A., Ural'tseva N.N., "Linear and quasilinear elliptic equations", Academic Press, New York and London, 1968.
- 243. Landis E.M., "The second order equations of elliptic and parabolic type", (in Russian) Nauka, Moscow, 1971. Engl. transl.: Transl. of Mathematical Monographs 171, AMS publications, 1998.
- 244. Lang S., "Differential and Riemannian manifolds", Graduate Texts in Mathematics 160, Springer-Verlag, New York, 1995.
- 245. Lax P.D, On the Cauchy problem for hyperbolic equations and the differentiability of solutions of elliptic equations, *Comm. Pure Appl. Math.*, 8 (1955) 615-633.
- 246. Li P., Uniqueness of L¹-solutions for the Laplace equation and the heat equation on Riemannian manifolds, J. Diff. Geom., 20 (1984) no.2, 447-457.
- 247. Li P., Curvature and function theory on Riemannian manifolds, Surveys in Diff. Geom., VII (2000) 375-432.
- 248. Li P., Schoen R., L^p and mean value properties of subharmonic functions on Riemannian manifolds, Acta Math., 153 (1984) 279-301.
- Li P., Schoen R., Yau S.-T., On the isoperimetric inequality for minimal surfaces, Ann. Scuola Norm. Sup. Pisa, 11 (1984) 237-244.
- 250. Li P., Tam L.F., Positive harmonic functions on complete manifolds with nonnegative curvature outside a compact set, Ann. Math., 125 (1987) 171–207.

- 251. Li P., Tam L.F., Harmonic functions and the structure of complete manifolds, J. Diff. Geom., 35 (1992) 359–383.
- Li P., Tam L.F., Green's function, harmonic functions and volume comparison, J. Diff. Geom., 41 (1995) 277–318.
- 253. Li P., Treibergs A., Applications of eigenvalue techniques to geometry, *in:* "Contemporary Geometry", Univ. Ser. Math., Plenum, New York, 1991. 21-52.
- 254. Li P., Wang J., Convex hull properties of harmonic maps, J. Diff. Geom., 48 (1998) 497-530.
- 255. Li P., Wang J., Mean value inequalities, Indiana Univ. Math. J., 48 (1999) 1257-1283.
- 256. Li P., Yau S.-T., Eigenvalues of a compact Riemannian manifold, Proc. Symp. Pure Math., 36 (1980) 205–239.
- 257. Li P., Yau S.-T., On the Schrödinger equation and the eigenvalue problem, Comm. Math. Phys., 88 (1983) 309–318.
- 258. Li P., Yau S.-T., On the parabolic kernel of the Schrödinger operator, Acta Math.,
 156 (1986) no.3-4, 153-201.
- 259. Liskevich V., Semenov Yu., Two-sided estimates of the heat kernel of the Schrödinger operator, Bull. London Math. Soc., 30 (1998) 596-602.
- Liskevich V., Semenov Yu., Estimates for fundamental solutions of second-order parabolic equations, J. London Math. Soc. (2), 62 (2000) no.2, 521–543.
- 261. Losev A.G., Some Liouville theorems on Riemannian manifolds of a special type, (in Russian) Izv. Vyssh. Uchebn. Zaved. Matematika, (1991) no.12, 15-24. Engl. transl.: Soviet Math. (Iz. VUZ), 35 (1991) no.12, 15-23.
- Losev A.G., Some Liouville theorems on non-compact Riemannian manifolds, (in Russian) Sibirsk. Mat. Zh., 39 (1998) no.1, 87-93. Engl. transl.: Siberian Math. J., (39) (1998) no.1, 74–80.
- 263. Lusternik L.A., Sobolev S.L., "Elements of functional analysis", Hindustan Publishing Corp., Delhi, 1974.
- 264. Lyons T., Instability of the Liouville property for quasi-isometric Riemannian manifolds and reversible Markov chains, J. Diff. Geom., 26 (1987) 33-66.
- 265. Lyons T., Random thoughts on reversible potential theory, in: "Summer School in Potential Theory, Joensuu 1990", Ed. Ilpo Laine, Publications in Sciences 26, University of Joensuu, 71-114.
- 266. Lyons T., Instability of the conservative property under quasi-isometries, J. Diff. Geom., 34 (1991) 483-489.
- 267. Malliavin P., Stroock D.W., Short time behavior of the heat kernel and its logarithmic derivatives, J. Diff. Geom., 44 (1996) 550-570.
- 268. Maz'ya V.G., On certain intergal inequalities for functions of many variables, (in Russian) Problemy Matematicheskogo Analiza, Leningrad. Univer., 3 (1972) 33-68. Engl. transl.: J. Soviet Math., 1 (1973) 205-234.
- 269. Maz'ya V.G., "Sobolev spaces", Springer, 1985.
- 270. Maz'ya V.G., Classes of domains, measures and capacities in the theory of differentiable functions, in: "Analysis, III", Encyclopaedia Math. Sci. 26, Springer, Berlin, 1991. 141-211.
- 271. McKean H.P., "Stochastic integrals", Academic Press, 1969.
- 272. McKean H.P., An upper bound to the spectrum of Δ on a manifold of negative curvature, J. Diff. Geom., 4 (1970) 359–366.
- 273. McOwen R., "Partial differential equations: methods and applications", Prentice Hall, 1996.
- 274. Milnor J., On deciding whether a surface is parabolic or hyperbolic, Amer. Math. Monthly, 84 (1977) 43-46.
- 275. Minakshisundaram S., Eigenfunctions on Riemannian manifolds, J. Indian Math. Soc., 17 (1953) 158-165.

- 276. Minakshisundaram S., Pleijel, A., Some properties of eigenfunctions of the Laplace operator on Riemannian manifolds, *Canad. J. Math.*, 1 (1949) 242-256.
- 277. Molchanov S.A., Diffusion processes and Riemannian geometry, (in Russian) Uspekhi Matem. Nauk, 30 (1975) no.1, 3-59. Engl. transl.: Russian Math. Surveys, 30 (1975) no.1, 1-63.
- 278. Morgan F., Isoperimetric estimates in products, Ann. Global Anal. Geom., 30 (2006) no.1, 73–79.
- 279. Moser J., On Harnack's theorem for elliptic differential equations, Comm. Pure Appl. Math., 14 (1961) 577-591.
- Moser J., A Harnack inequality for parabolic differential equations, Comm. Pure Appl. Math., 17 (1964) 101-134. Correction: Comm. Pure Appl. Math., 20 (1967) 231-236.
- 281. Mukherjea A., Pothoven K., "Real and functional analysis. Part B. Functional analysis", Plenum Press, New York, 1986.
- Murata M., Positive solutions and large time behaviors of Schrödinger semigroups, Simon's problem, J. Funct. Anal., 56 (1984) 300-310.
- 283. Murata M., Large time asymptotics for fundamental solutions of diffusion equations, *Tôhoku Math. J.*, 37 (1985) 151-195.
- 284. Murata M., Structure of positive solutions to $(-\Delta + V)u = 0$ in \mathbb{R}^n , Duke Math. J., 53 (1986) no.4, 869-943.
- 285. Murata M., Positive harmonic functions on rotationary symmetric Riemannian manifolds, *in:* "Potential Theory", Ed. M.Kishi, Walter de Gruyter, Berlin, 1992. 251-259.
- 286. Murata M., Uniqueness and nonuniqueness of the positive Cauchy problem for the heat equation on Riemannian manifolds, *Proceedings of AMS*, **123** (1995) no.6, 1923-1932.
- 287. Mustapha S., Gaussian estimates for heat kernels on Lie groups, Math. Proc. Camb. Phil. Soc., 128 (2000) 45-64.
- 288. Nadirashvili N.S., A theorem of Liouville type on a Riemannian manifold, (in Russian) Uspekhi Matem. Nauk, 40 (1985) no.5, 259-260. Engl. transl.: Russian Math. Surveys, 40 (1986) no.5, 235-236.
- Nadirashvili N.S., Metric properties of eigenfunctions of the Laplace operator on manifolds, Ann. Inst. Fourier (Grenoble), 41 (1991) no.1, 259–265.
- Nadirashvili N.S., Isoperimetric inequality for the second eigenvalue of a sphere, J. Diff. Geom., 61 (2002) no.2, 335–340.
- 291. Nadirashvili N.S., Toth J.A., Jakobson D., Geometric properties of eigenfunctions, Uspekhi Mat. Nauk, 56 (2001) no.6, 67–88. (in Russian) Engl. transl.: Russian Math. Surveys, 56 (2001) no.6, 1085–1105.
- 292. Nash J., Continuity of solutions of parabolic and elliptic equations, Amer. J. Math., 80 (1958) 931-954.
- 293. Nirenberg L., Remarks on strongly elliptic partial differential equations, Comm. Pure Appl. Math., 8 (1955) 649-675.
- 294. Nirenberg L., On elliptic partial differential equations, Ann. Scuola Norm. Sup. Pisa, 13 (1959) 115-162.
- 295. Norris J.R., Heat kernel bounds and homogenization of elliptic operators, Bull. London Math. Soc., 26 (1994) 75-87.
- 296. Norris J.R., Heat kernel asymptotics and the distance function in Lipschitz Riemannian manifolds, Acta Math., 179 (1997) 79-103.
- 297. Oleinik O.A., Radkevich E.V., "Second order equations with nonnegative characteristic form", AMS, Plenum Press, 1973.
- Oleinik O.A., Radkevich E.V., The method of introducing a parameter in the study of evolutionary equations, (in Russian) Uspekhi Matem. Nauk, 33 (1978) no.5, 7-76. Engl. transl.: Russian Math. Surveys, 33 (1978) no.5, 7-84.

- 299. Petersen P., "Riemannian geometry", Graduate Texts in Mathematics 171, Springer, 1998.
- 300. Pinchover Y., Large time behavior of the heat kernel and the behavior of the Green function near criticality for nonsymmetric elliptic operators, J. Funct. Anal., 104 (1992) no.1, 54-70.
- 301. **Pinchover Y.**, On non-existence of any λ_0 -invariant positive harmonic function, a counter example to Stroock's conjecture, Comm. Partial Differential Equations, **20** (1995) 1831-1846.
- 302. Pittet Ch., The isoperimetric profile of homogeneous Riemannian manifolds, J. Diff. Geom., 54 (2000) no.2, 255-302.
- 303. **Pittet Ch., Saloff-Coste L.**, A survey on the relationship between volume growth, isoperimetry, and the behavior of simple random walk on Cayley graphs, with examples, preprint
- 304. Pittet Ch., Saloff-Coste L., On the stability of the behavior of random walks on groups, J. Geom. Anal., 10 (2000) no.4, 713-737.
- 305. Pólya G., Szegö G., "Isoperimetric inequalities in mathematical physics", Princeton University Press, Princeton, 1951.
- 306. Porper F.O., Eidel'man S.D., Two-side estimates of fundamental solutions of second-order parabolic equations and some applications, (in Russian) Uspekhi Matem. Nauk, **39** (1984) no.3, 101-156. Engl. transl.: Russian Math. Surveys, **39** (1984) no.3, 119-178.
- 307. Qian Z., Gradient estimates and heat kernel estimates, Proc. Roy. Soc. Edinburgh, 125A (1995) 975-990.
- 308. Rauch J., "Partial differential equaions", Graduate Texts in Mathematics 128, Splinger-Verlag, 1991.
- Reed M., Simon B., "Methods of modern mathematical physics. II: Fourier analysis, self-adjointness", Academic Press, 1975.
- Riesz F., Sz.-Nagy B., "Functional analysis", Dover Publications, Inc., New York, 1990.
- 311. Robin L., "Fonctions sphériques de Legendre et fonctions sphéroïdales. Tome III", Collection Technique et Scientifique du C. N. E. T. Gauthier-Villars, Paris, 1959.
- Robinson D.W., "Elliptic operators and Lie groups", Oxford Math. Mono., Clarenton Press, Oxford New York Tokyo, 1991.
- 313. Röckner M., Wang F.Y., Weak Poincaré inequalities and L²-convergence rates of Markov semigroups, J. Funct. Anal., 185 (2001) 564-603.
- 314. Röckner M., Wang F.Y., Harnack and functional inequalities for generalized Mehler semigroups, J. Funct. Anal., 203 (2003) 237-261.
- 315. Röckner M., Wang F.Y., Supercontractivity and ultracontractivity for (nonsymmetric) diffusion semigroups on manifolds, *Forum Math.*, 15 (2003) no.6, 893–921.
- 316. Roelcke W., Über den Laplace-Operator auf Riemannschen Mannigfaltigkeiten mit diskontinuierlichen Gruppen, Math. Nachr., 21 (1960) 131–149.
- 317. Rosenberg S., "The Laplacian on a Riemannian manifold", London Mathematical Society Student Texts 31, Cambridge University Press, 1997.
- 318. Rudin W., "Functional analysis", McGraw-Hill, Inc., New York, 1991.
- Saloff-Coste L., A note on Poincaré, Sobolev, and Harnack inequalities, Internat. Math. Res. Notices, 2 (1992) 27-38.
- 320. Saloff-Coste L., Uniformly elliptic operators on Riemannian manifolds, J. Diff. Geom., 36 (1992) 417-450.
- 321. Saloff-Coste L., Isoperimetric inequalities and decay of iterated kernels for almosttransitive Markov chains, Combinatorics, Probability, Computing, 4 (1995) no.4, 419–442.

- 322. Saloff-Coste L., Parabolic Harnack inequality for divergence form second order differential operators, *Potential Analysis*, 4 (1995) 429-467.
- 323. Saloff-Coste L., Lectures on finite Markov chains, *in:* "Lectures on probability theory and statistics", Lecture Notes Math. 1665, Springer, 1997. 301-413.
- 324. Saloff-Coste L., "Aspects of Sobolev inequalities", LMS Lecture Notes Series 289, Cambridge Univ. Press, 2002.
- 325. Saloff-Coste L., Pseudo-Poincaré inequalities and applications to Sobolev inequalities, preprint
- 326. Schoen R., Yau S.-T., "Lectures on Differential Geometry", Conference Proceedings and Lecture Notes in Geometry and Topology 1, International Press, 1994.
- 327. Schwartz L., "Théorie des distributions. I, II", Act. Sci. Ind. 1091, 1122, Hermann et Cie., Paris, 1951.
- Sobolev S.L., On a theorem of functional analysis, (in Russian) Matem. Sbornik,
 4 (1938) 471-497.
- Sternberg S., "Lectures on differential geometry", Chelsea Publishing Co., New York, 1983.
- 330. Strichartz R.S., Analysis of the Laplacian on the complete Riemannian manifold, J. Funct. Anal., 52 (1983) no.1, 48-79.
- 331. Strichartz R.S., Laplacians on fractals with spectral gaps have nicer Fourier series, Math. Res. Lett., 12 (2005) no.2-3, 269–274.
- 332. Strichartz R.S., "Differential equations on fractals. A tutorial", Princeton University Press, Princeton, NJ, 2006.
- 333. Stroock D.W., Estimates on the heat kernel for the second order divergence form operators, in: "Probability theory. Proceedings of the 1989 Singapore Probability Conference held at the National University of Singapore, June 8-16 1989", Ed. L.H.Y. Chen, K.P. Choi, K. Hu and J.H. Lou, Walter De Gruyter, 1992. 29-44.
- Stroock D.W., "Probability Theory. An analytic view", Cambridge Univ. Press, 1993.
- 335. Stroock D.W., "A concise introduction to the theory of integration", Birkhäuser Boston, Inc., Boston, MA, (1999)
- 336. Stroock D.W., Turetsky J., Upper bounds on derivatives of the logarithm of the heat kernel, Comm. Anal. Geom., 6 (1998) no.4, pa669–685
- 337. Sturm K-Th., Heat kernel bounds on manifolds, Math. Ann., 292 (1992) 149-162.
- 338. Sturm K-Th., Analysis on local Dirichlet spaces I. Recurrence, conservativeness and L^p-Liouville properties, J. Reine. Angew. Math., 456 (1994) 173-196.
- 339. Sturm K-Th., Analysis on local Dirichlet spaces II. Upper Gaussian estimates for the fundamental-solutions of parabolic equations, Osaka J. Math, 32 (1995) no.2, 275-312.
- 340. Sullivan D., Related aspects of positivity in Riemannian geometry, J. Diff. Geom.,
 25 (1987) 327-351.
- 341. Sung C.-J., Tam L.-F., Wang J., Spaces of harmonic functions, J. London Math. Soc. (2), (2000) no.3, 789–806.
- 342. Täcklind S., Sur les classes quasianalytiques des solutions des équations aux dérivées partielles du type parabolique, Nova Acta Regalis Societatis Scientiarum Uppsaliensis, (4), 10 (1936) no.3, 3-55.
- 343. Takeda M., On a martingale method for symmetric diffusion process and its applications, Osaka J. Math, 26 (1989) 605-623.
- 344. Takeda M., Subcriticality and conditional gaugeability of generalized Schrödinger operators, J. Funct. Anal., 191 (2002) no.2, 343–376.
- 345. Takeda M., Gaussian bounds of heat kernels of Schrödinger operators on Riemannian manifolds, Bull. London Math. Soc., 39 (2007) 85-94.

- 346. Taylor A.E., Lay D.C., "Introduction to functional analysis", Robert E. Krieger Publishing Co., Inc., Melbourne, FL, 1986.
- 347. Tikhonov A.N., Uniqueness theorems for the equation of heat conduction, (in Russian) Matem. Sbornik, 42 (1935) 199-215.
- 348. **Treves F.**, "Introduction to pseudodifferential and Fourier integral operators, I", Plenum Press, 1982.
- 349. Urakawa H., Geometry of Laplace-Beltrami operator on a complete Riemannian manifold, in: "Progress in differential geometry", Advanced Studies in Pure Math.22, Math. Soc. Japan, Tokyo, (1993) 347-406.
- 350. Ushakov V.I., Stabilization of solutions of the third mixed problem for a second order parabolic equation in a non-cylindric domain, (in Russian) Matem. Sbornik, 111 (1980) 95-115. Engl. transl.: Math. USSR Sb., 39 (1981) 87-105.
- 351. Varadhan S.R.S., On the behavior of the fundamental solution of the heat equation with variable coefficients, *Comm. Pure Appl. Math.*, **20** (1967) 431-455.
- 352. Varopoulos N.Th., Potential theory and diffusion of Riemannian manifolds, in: "Conference on Harmonic Analysis in honor of Antoni Zygmund. Vol I, II.", Wadsworth Math. Ser., Wadsworth, Belmont, Calif., 1983. 821-837.
- 353. Varopoulos N.Th., Hardy-Littlewood theory for semigroups, J. Funct. Anal., 63 (1985) no.2, 240-260.
- 354. Varopoulos N.Th., Isoperimetric inequalities and Markov chains, J. Funct. Anal.,
 63 (1985) no.2, 215-239.
- 355. Varopoulos N.Th., Saloff-Coste L., Coulhon T., "Analysis and geometry on groups", Cambridge Tracts in Mathematics, 100, Cambridge University Press, Cambridge, 1992.
- 356. Vladimirov V. S., "Methods of the theory of generalized functions", Analytical Methods and Special Functions 6, Taylor & Francis, London, 2002.
- 357. Weyl H., The method of orthogonal projection in potential theory, Duke Math. J., 1 (1940) 414-444.
- 358. Woess W., Random walks on infinite graphs and groups a survey on selected topics, Bull. LMS, 26 (1994) 1-60.
- 359. Woess W., "Random walks on infinite graphs and groups", Cambridge Tracts in Mathematics 138, Cambridge Univ. Press., 2000.
- 360. Yan J.-A., A formula for densities of transition functions, in: "Sem. Prob. XXII", 92-100. Lecture Notes Math. 1321,
- Yau S.-T., Harmonic functions on complete Riemannian manifolds, Comm. Pure Appl. Math., 28 (1975) 201-228.
- 362. Yau S.-T., Isoperimetric constant and the first eigenvalue of a compact Riemannian manifold, Ann. Sci. Ecole Norm. Sup., 4th serie, 8 (1975) no.4, 487-507.
- 363. Yau S.-T., Some function-theoretic properties of complete Riemannian manifolds and their applications to geometry, *Indiana Math. J.*, **25** (1976) 659-670.
- 364. Yau S.-T., On the heat kernel of a complete Riemannian manifold, J. Math. Pures Appl., ser. 9, 57 (1978) 191-201.
- 365. Yau S.-T., Survey on partial differential equations in differential geometry, Ann. Math. Studies, 102 (1982) 3-70.
- 366. Yau S.-T., Harnack inequality for non-self-adjoint evolution equations, Math. Research Letters, 2 (1995) 387-399.
- 367. Yau S.-T., An estimate of the gap of the first two eigenvalues in the Schrödinger operator, in: "Lectures on partial differential equations", New Stud. Adv. Math. 2, Int. Press, Somerville, MA, 2003. 223–235.
- 368. Yosida K., "Functional analysis", Springer, 1980.
- 369. Zegarlin'ski B., Recent progress in the hypercontractive semigroups, in: "Seminar on Stochastic Analysis, Random Fields and Applications (Ascona, 1993)", Progr. Probab. 36, Birkhäuser, Basel, 1995. 253–262.

- 370. Zhang Q.S., Large time behavior of Schrodinger heat kernels and applications, Comm. Math. Phys., 210 (2000) no.2, 371–398.
- 371. Zhang Q.S., Zhao Z., Estimates of global bounds for some Schrödinger heat kernels on manifolds, Illinois J. Math., 44 (2000) no.3, 556–573.

Some notation

- $\mathbb{R}_+ \equiv (0, +\infty)$
- $\bullet~{\rm esup}$ the essential supremum
- $\bullet~{\rm einf}$ the essential infimum
- $f_+ \equiv \frac{1}{2}(|f|+f)$ the positive part
- $f_{-} \equiv \frac{1}{2}(|f| f)$ the negative part
- $[f]_a^b \equiv f(b) f(a)$
- $\log_+ x \equiv (\log x)_+$
- \simeq "comparable to"; namely, $f(x) \simeq g(x)$ if there exists a constant C > 0 such that $C^{-1}g(x) \le f(x) \le Cg(x)$ for all x from a specified domain.
- $B_r(x)$ a ball in \mathbb{R}^n , that is, $B_r(x) = \{y \in \mathbb{R}^n : |x y| < r\}$
- $B_r \equiv B_r(0) = \{ y \in \mathbb{R}^n : |y| < r \}.$
- ω_n the area of the unit (n-1)-sphere in \mathbb{R}^n .
- 1_A the indicator function of a set A, that is, $1_A(x) = 1$ if $x \in A$ and $1_A(x) = 0$ otherwise.
- \Subset "compact inclusion"; $A \Subset B$ means that the closure \overline{A} of the set A is compact and $\overline{A} \subset B$.
- \mathcal{H} a Hilbert space
- \rightarrow the sign of the weak convergence (in a Hilbert space)
- M a Riemannian manifold
- $x \to \infty$ a sequence of points on a manifold eventually leaving any compact set.
- \mathbf{g} the Riemannian metric on M
- μ a reference measure on M
- Δ_{μ} the weighted Laplace operator on M
- \mathcal{L} the Dirichlet Laplace operator on M
- $L^{p}(M,\mu)$ the Lebesgue function space
- $\|\cdot\|_p \equiv \|\cdot\|_{L^p}$
- $u \leq v \mod W_0^1$ means that $u \leq v + w$ where $w \in W_0^1$.
- d the geodesic distance on M
- B(x, r) a geodesic ball on M with respect to the geodesic distance d(x, y).

Conventions.

• Summation is assumed over repeated indices. For example,

$$\xi_i x^i = \sum_{i=1}^n \xi_i x^i, \ a^{ij} u_j = \sum_{j=1}^n a^{ij} u_j, \ g_{ij} v^i v^j = \sum_{i,j=1}^n g_{ij} v^i v^j,$$

etc.

- Letters c, C, c', C', etc denote positive constants (depending on specified parameters) whose value may change at each occurrence.
- positive=strictly positive, negative=strictly negative, decreasing=non-increasing=non-decreasing

476

Index

 $|\alpha|$ - order of multiindex, 15 $\left[\alpha\right]$ - the weighted order, 170 $\mathcal{B}(M)$ - the class of Borel measurable functions on M, 59 B(x,r) - the geodesic ball, 89 $B_r(x)$ - the Euclidean ball, 17 $C(\Omega), 15$ $C_b(M), 202$ $C_{b}(\Omega), 120$ $C_b(\mathbb{R}^n), 40$ $C^{k}(M), 51$ $C^k(\Omega), 15$ $C_0^k(M), 51$ $C_{h}^{k}(\mathbb{R}^{n}), 41$ $C^{\infty}(\Omega), 16$ $C_0^{\infty}(\Omega), 16$ \hookrightarrow , 16 €, 16, 49 $\simeq, 36, 93$ $\xrightarrow{C^{\infty}}$, 186 $\xrightarrow{\mathcal{D}}$, 24, 97 $\xrightarrow{\mathcal{D}'}$, 24, 97 $\stackrel{\mathcal{W}_{loc}^{\infty}}{\longrightarrow},\,186$ $\mathcal{D}'(M), 97$ $\mathcal{D}'(\Omega), 24$ $\mathcal{D}(M), 97$ $\mathcal{D}(\Omega), 23$ $\vec{\mathcal{D}}'(M), 98$ $\vec{\mathcal{D}}(M), 98$ δ_i^i - the Kroneker delta, 56 Δ - the Laplace operator, 1 Δ_{μ} - the weighted Laplace operator, 68 diag, 260 d(x, y) - the geodesic distance, 86 dJ - the tangent map, 92 $E_D(t, x), 399$ E_U - the spectral measure of U, 266

 E_{λ} - a spectral resolution, 111, 113, 132, 190, 266, 449, 452 f * g - convolution, 7, 17 φ_{ε} - mollifier, 18 G - the Green operator, 341 g(x, y) - the Green function, 342 $g^{\Omega}(x,y)$ - the Green function in Ω , 342 g - a Riemannian metric, 56 $g^{-1}, 58$ $\mathbf{g}_{\mathbb{H}^n}$ - the canonical metric on \mathbb{H}^n , 77 $\mathbf{g}_{\mathbb{R}^n}$ - the canonical metric on \mathbb{R}^n , 57 $\mathbf{g}_{\mathbb{S}^n}$ - the canonical metric on \mathbb{S}^n , 72 Γ -function, 455 Γ - a function class, 372 Γ_{δ} - a function class, 376 Γ_{δ} - a function class, 376 Gf, 341 $G^{\Omega}f, 341$ $g_{ij}, 57$ $g^{ij},\,58$ $\langle \cdot, \cdot \rangle$ - inner product of tangent vectors, 57 $\langle \cdot, \cdot \rangle$ - pairing of vectors and covectors, 56 $(f,g)_{L^2}$ - the inner product in $L^2,\,440$ J_* - pullback operator, 92 \mathcal{L} - the Dirichlet Laplace operator, 105 \mathcal{L}^{Ω} - the Dirichlet Laplace operator in Ω , 144 Λ - a Faber-Krahn function, 367 $\Lambda(M)$ - the class of Lebesgue measurable functions on M, 59 $\lambda_k(\Omega), 277$ $\lambda_{\min}(A), 265$ $\lambda_{\min}(M), 111, 271$ L - a function class, 371 \mathbf{L}_{δ} - a function class, 376 $\widetilde{\mathbf{L}}_{\delta}$ - a function class, 376 $\ell(\gamma)$ - the length of a path, 86

Lip(M), 296

 $Lip_{0}(M), 299$ $Lip_{loc}(M), 299$ $\log_{+}, 378$ $L^{p}(M), 98, 440$ $L_{loc}^{p}(M), 98$ $L^{p}(\Omega), 16$ $\vec{L}^{p}(M), 98$ $\vec{L}_{loc}^{p}(M), 99$ m(U), 267 ∇ , 6, 43, 58 $\|\cdot\|_{C^k}, 16$ $\|\cdot\|_{L^p}, 440$ $\|\cdot\|_{p}, 439$ $\|\cdot\|_{V^k}, k < 0, 171$ $\|\cdot\|_{V^k}, k \ge 0, 170$ $\|\cdot\|_{W^1}, 100$ $\|\cdot\|_{W^k}, k < 0, 37$ $\|\cdot\|_{W^k}, \ k \ge 0, \ 34$ $\|\cdot\|_{\mathcal{W}^{2k}}, 183$ $\|\cdot\|_{p\to q}, 365$ $\|\cdot\|_{Lip}, 296$ P_t - the heat semigroup $=e^{-t\mathcal{L}}, 115, 117, 130$ a convolution operator, 40 a smooth version of $e^{-t\mathcal{L}}$, 191 an integral operator, 201 P_t^{Ω} - the heat semigroup in Ω , 144 $p_t(x)$ - the heat kernel in \mathbb{R}^n , 4 $p_t(x, y)$ - the heat kernel, 198 $p_{t,x}(y), 191$ $\mathcal{R}(f)$ - the Rayleigh quotient, 272 R_{α} - the resolvent, 106, 130, 219 R^{Ω}_{α} - the resolvent in Ω , 144 $r_{\alpha}(x,y)$ - the resolvent kernel, 262 R^k - the iterated resolvent, 133 supp - support of a continuous function, 3, 51 of a distribution, 26, 97 of a function from L_{loc}^1 , 98 $u = w \mod W_0^1(M), \ 135$ $u \leq w \mod W_0^1(M), 135$ | |, 61 V(x,r), 303, 409 $V^{k}(\Omega), \ k < 0, 171$ $V^k(\Omega), \ k \ge 0, 170$ $V_{loc}^k(\Omega), 171$ $W^{1}(M), 100$ $W_0^1(M), 104$ $W_{c}^{1}(M), 127$ $W_{loc}^{1}(M), 128$ $W^2(M), 104$ $W_0^2(M), 104$

 $W_{loc}^{2}(M), 130$ $W^{k}(\Omega), \ k < 0, \ 37$ $W^{k}(\Omega), k \geq 0, 34$ $W_0^1(\Omega), 36, 158$ $W^{\infty}(\Omega), 152$ $W_{loc}^{\infty}(\Omega), 152$ $W_{loc}^{k}\left(\Omega\right), \ k < 0, \ 38$ $W_{loc}^{k}\left(\Omega\right), \, k \geq 0, \, 34$ $\mathcal{W}_{0}^{s}(M), 188$ $\mathcal{W}^{2k}(M), 183$ $\mathcal{W}^{2k}_{loc}(M), 183$ $\mathcal{W}_{loc}^{\infty}\left(M\right),\,186$ $\omega_n, 3, 82, 83$ σ -Algebra, 435 Almost everywhere, 438 Anisotropic Sobolev spaces, 170 Area function, 82 Aronson, Donald G., 215, 339, 414 Atlas, 50 Azencott, Robert, 320 Basis in a Hilbert space, 432 Beltrami, Eugenio, ix Bessel semigroup, 133 Bessel's inequality, 432 Borel set in \mathbb{R}^n , 436 on a manifold, 59 Bottom eigenfunction, 275 Bottom of the spectrum, 265 Boukricha, Abderrahman, 362 Bounded convergence theorem, 114, 439 Bounded geometry, 312 Brooks, Robert, 320 C-manifold, 49 Canonical Euclidean metric, 57 Canonical hyperbolic metric, 77 Canonical spherical metric, 72 Carathéodory extension theorem, 435 Carlen, Eric A., 388 Carron, Gilles, 388 Cartan-Hadamard manifold, 368, 383 Cauchy problem, 4 in $L^2(\mathbb{R}^n)$, 45 L^2 -Cauchy problem, 112 Cauchy semigroup, 134 Cauchy-Schwarz inequality, 431, 440 Chain rule for Lipschitz functions, 301 for strong derivatives, 121 for the Riemannian gradient, 59

478

for the weighted Laplacian, 69 in W^1 , 128 in W^1_0 , 123, 124 Chart, 49 Chavel, Isaac, 388 Cheeger's inequality, 275 C^k -norm, 16 Closed operator, 109, 446 Compact embedding theorem, 214, 289 in \mathbb{R}^n , 158 Compact inclusion, 16, 49 Compact operator, 168, 434 Comparison principle, 137 Complete measure, 435 Completeness of L^p , 440 Components of a vector, 55 of the metric tensor, 57 Convergence in $\mathcal{D}(\Omega)$, 23 in $\mathcal{D}(M), 97$ Convex function, 42 Convexity lemma, 43 Convolution, 17 Cotangent space, 56 Coulhon, Thierry, 388 Countable base, 49 Counting measure, 267 Covector, 56 Cutoff function, 19 Lipschitz, 300 on a manifold, 52 Davies, Edward Brian, x, 339 Davies-Gaffney inequality, 326 De Broglie, Louis, 2 De Giorgi, Ennio, 181, 215, 414 Delta function, 24 Density function, 67 Density of measure, 438 Diffeomorphism, 92 Differential, 56 Dirac, Paul, 2 Dirichlet Laplace operator, 105 Dirichlet problem, 105 weak, 105, 111, 135 Discrete spectrum, 265 Distribution definition, 24 derivatives, 25 multiplication by a function, 25 non-negative, 136

on a manifold, 97

support, 26, 97 Distributional gradient, 99 Distributional vector field, 98 Divergence on a manifold, 64 weighted, 68 Divergence theorem in \mathbb{R}^n , 3 on a manifold, 64 Dodziuk, Józef, 263, 429 Dominated convergence theorem, 439, 441 Doob. 252 Doubling volume property, 410 Eigenvalue, 434 Eigenvector, 434 Einstein, Albert, 2 Elliptic operator, 4, 162 Ellipticity constant, 162 Embedding of linear topological spaces, 16 Essential spectrum, 265 Exhaustion sequence, 52, 144 compact, 52, 201 Faber-Krahn inequality, 367 in balls, 397 in unions of balls, 402 on direct products, 386 relative, 409 Faber-Krahn theorem, 367 Fatou's lemma, 438 Fourier series, 432 Fourier transform, 8 inversion formula, 155 Fourier, Jean Baptiste Joseph, 1 Friedrichs lemma, 160 Friedrichs, Kurt Otto, 181 Friedrichs-Poincaré inequality, 159 Fubini's theorem, 442 Functional calculus of operators, 453 Fundamental solution of the heat equation, 243 of the Laplace operator, 342, 359 regular, 243 Fundamental theorem of calculus, 120 Γ -transform, 372 Gaffney, Matthew P., ix, 319 Gamma function, 455 Gâteaux derivative, 210 Gauss-Weierstrass function, 4 Gaussian upper bounds, 391 Geodesic ball, 89

INDEX

Geodesic completeness, 295 Geodesic distance, 86 Geodesics, 86, 295 Gradient, 58 Green formula, 104 for Laplacian on a manifold, 67 in \mathbb{R}^n , 3 Green function, 342 upper bound, 414 Green operator, 341 Gross, Leonard, 388 Ground state, 358 Gushchin Anatolii Konstantinovich, 320 h-transform, 252 Hamilton, Richard, x Hansen, Wolfhard, 362 Hardy inequality, 259 Harmonic function, 83, 189, 229 α -Harmonic function, 229, 354, 356 Harnack inequality in \mathbb{R}^n , 355 local, 353 Harnack principle, 356 Hausdorff space, 49 Heat kernel asymptotics as $t \to \infty$, 292 existence, 191, 428 in half-space, 258 in \mathbb{H}^n , 256 in \mathbb{R}^n , 4 in Weyl's chamber, 258 integrated upper bound, 399, 422 Li-Yau upper estimate, 413 of a weighted manifold, 198 off-diagonal upper bound, 404, 410 on model manifolds, 251 on products, 249 on-diagonal lower bound, 424 on-diagonal upper bound, 380 smoothness, 198, 208 under change of measure, 252 under isometry, 250 Heat semigroup in \mathbb{R}^n , 40 on a manifold, 115 Hermite polynomials, 69 Hilbert space, 431 Hilbert-Schmidt theorem, 434 Hölder conjugate, 439 Hölder inequality, 439 Hopf-Rinow Theorem, 295, 296 Hörmander, Lars Valter, 181

Hyperbolic space, 77 Induced measure, 71 Induced metric, 71 Infinity point ∞ on a manifold, 141 Initial value problem, 4 Integrable function, 437 Integral maximum principle, 321 Integration by parts formula, 3 Isometric manifolds, 92 Jacobian matrix, 60 Jorgenson, Jay, ix Khas'minskii, Rafail Zalmanovich, 241 Krylov, Nikolai Vladimirovich, 181 Kusuoka, Shigeo, 388 L-transform, 372 Landis, Evgeniy Mikhailovich, 182 Lang, Serge, ix Laplace equation, 1 Laplace operator Dirichlet, 105 distributional, 99 in \mathbb{R}^n , 1 on a manifold, 67 weak, 99 weighted, 68 Laplace, Pierre-Simon, 1 Lax, Peter David, 181 Lebesgue integral, 437 Lebesgue integral sum, 437 Lebesgue measure, 436 Lebesgue space, 440 in \mathbb{R}^n , 16 in \mathbb{R}^n , local, 16 Length of a path, 86 Levy distribution, 134 Li, Peter, 415 Li-Yau estimate, 413 Liouville theorem, 355 Lipschitz constant, 33, 296 Lipschitz function, 33, 296 Local coordinate system, 49 Locally Lipschitz function, 299 Log-convex function, 43 Lyons, Terry, 320, 339 Markovian properties, 123 Maximum/minimum principle elliptic, 230, 293 elliptic, exterior, 189 elliptic, in \mathbb{R}^n , 13

480

for superaveraging functions, 350 parabolic, 223 parabolic, in \mathbb{R}^n , 9 strong, elliptic, 229 strong, for superaveraging functions, 350 strong, parabolic, 225, 230 weak, elliptic, 136 weak, parabolic, 138, 141 Maxwell, James Clerk, 2 Maz'ya, Vladimir Gilelevich, 389 McKean, Henry P., Jr., 263 Mean value inequality, 391 Measurable function, 436 Measurable set, 436 on a manifold, 59 Measure abstract, 434 σ -finite, 434 Measure space, 437 Mehler kernel, 255, 263, 303 Minakshisundaram S., ix Minkowski metric, 77 Model manifold with two ends, 240 Mollifier, 18 Monotone convergence theorem, 439 Moser inequality, 371 Moser, Jürgen K., 215, 414 Multiindex, 15, 170 order of, 15 weighted order of, 170 Nash inequality, 368 generalized, 368 Nash, John Forbes, 181, 215, 388 r-Neighborhood, 324 Nirenberg, Louis, 181 Null set, 435 Oleinik, Olga Arsen'evna, 320 One point compactification, 141 Parabolic manifold, 313 Parabolic operator, 4, 172 Parseval identity, 432 Partition of unity in \mathbb{R}^n , 19 on a manifold, 52 Perel'man, Grigory Ya., x Polar coordinates in \mathbb{H}^n , 77 in \mathbb{R}^n , 74 in \mathbb{S}^n , 75 on a model manifold, 80

Positive spectrum, 357 Principle of uniform boundedness, 211, 433Product measure, 442 Product rule, 53, 59, 69 for distributional derivatives, 33 of higher order, 28 for Lipschitz functions, 300 for strong derivatives, 120 for the distributional gradient, 101 in W_0^1 , 111 Projector, 431 Pullback, 91 Push forward, 92 Push forward measure, 94 Quasi-isometric manifolds, 93 Quasi-isometry, 312 \mathbb{R} -differentiation, 53 Radkevich, Evgenii Vladimirovich, 320 Radon-Nikodym derivative, 438 Rayleigh quotient, 272 Regular measure, 436 Relative Faber-Krahn inequality, 409 Rellich theorem. 158 Resolvent, 106, 167, 219 Riemannian manifold, 57 complete, 295 Riemannian measure, 59 Riemannian metric tensor, 56 Riemannian model, 80 Riesz Representation Theorem, 431 Safonov, Mikhail V., 182 Saloff-Coste, Laurent, 389 Schrödinger, Erwin, 2 Schwartz, Laurent-Moïse, 181 Self-adjoint operator, 434 Simple function, 437 Smooth manifold, 50 Sobolev embedding theorem, 151, 214 Sobolev spaces in \mathbb{R}^{n} , 34, 37 in \mathbb{R}^{n+1} , anisotropic, 170 local, 34 on manifolds, 104 Sobolev, Sergei Lvovich, 181 Spectral mapping theorem, 453 Spectral theorem, 452 Stochastic completeness, 231 Stone-Weierstrass theorem, 283 Strichartz, Robert Stephen, 215, 319 Strong derivative, 45

Strong topology, 432 Strongly differentiable, 45 Stroock, Daniel W., 388 Subharmonic function, 229 α -Subharmonic function, 229 Submanifold, 70 Subsolution, 391 Sullivan, Dennis Parnell, 363 sup-norm, 16 Superaveraging function, 348 Superharmonic function, 229 α -Superharmonic function, 229 Supersolution, 217 Symmetric operator, 168 Täcklind class, 305 Takeda's inequality, 338 Takeda, Masayoshi, 339 Tangent space, 53 Tangent vector, 53 Test function, 23 Tikhonov class, 305 Tikhonov theorem, 12 Tikhonov, Andrey Nikolayevich, 320 Tonelli's theorem, 443 Transmutation formula, 121 Ultracontractive semigroup, 365 Ushakov, Vladimir Ignat'evich, 430 Varadhan, Srinivasa R. S., 414 Varopoulos, Nickolas Th., x, 388 Vector field, 56 Volume function, 82 Wave equation, 121, 197 finite propagation speed, 327 Wave operators, 121 Weak compactness, 433 Weak compactness of balls, 433 Weak convergence, 432 Weak derivative, 34 Weak gradient, 99 Weak topology, 432 in \mathcal{D}' , 24 Weighted manifold, 67 Weighted model, 82 Weyl's lemma, 181 Weyl, Hermann Klaus Hugo, 181 Yau, Shing-Tung, x, 319, 320, 363, 415

482

Titles in This Series

- 47 Alexander Grigor'yan, Heat Kernel and Analysis on Manifolds, 2009
- 46.2 Kenji Fukaya, Yong-Geun Oh, Hiroshi Ohta, and Kaoru Ono, Lagrangian Intersection Floer Theory, 2009
- 46.1 Kenji Fukaya, Yong-Geun Oh, Hiroshi Ohta, and Kaoru Ono, Lagrangian Intersection Floer Theory, 2009
 - 45 Lydia Bieri and Nina Zipser, Extensions of the Stability Theorem of the Minkowski Space in General Relativity, 2009
 - 44 Eric Sharpe and Arthur Greenspoon, Editors, Advances in String Theory, 2008
 - 43 Lizhen Ji, Editor, Arithmetic Groups and Their Generalizations, 2008
- 42.1 Ka-Sing Lau, Zhou-Ping Xin, and Shing-Tung Yau, Editors, Third International Congress of Chinese Mathematicians, 2008
 - 41 Wen-Ching Winnie Li, Editor, Recent Trends in Coding Theory and its Applications, 2007
 - 40 **Ovidiu Caliin, Der-Chen Chang, and Peter Greiner, Editors,** Geometric Analysis on the Heisenberg Group and Its Generalizations, 2007
 - 39 Zhijie Chen, Sheng-Li Tan, Jianpan Wang, and Stephen S.-T. Yau, Editors, Proceedings of the International Conference on Complex Geometry and Related Fields, 2007
 - 38 Noriko Yui, Shing-Tung Yau, and James D. Lewis, Editors, Mirror Symmetry V, 2006
 - 37 Lizhen Ji, Jian-Shu Li, H. W. Xu, and Shing-Tung Yau, Editors, Lie Groups and Automorphic Forms, 2006
 - 36 Chuu-Lian Terng, Editor, Integrable Systems, Geometry, and Topology, 2006
 - 35 Felix Finster, The Principle of the Fermionic Projector, 2006
 - 34 Ren-Hong Wong, Editor, Computational Geometry, 2003
 - 33 Eric D'Hoker, Duong Phong, and Shing-Tung Yau, Mirror Symmetry IV, 2002
 - 32 Xi-Ping Zhu, Lectures on Mean Curvature Flows, 2002
 - 31 Kiyoshi Igusa, Higher Franz-Reidemeister Torsion, 2002
 - 30 Weiman Han and Mircea Sofonea, Quasistatic Contact Problems in Viscoelasticity and Viscoelasticity, 2002
 - 29 S. T. Yau and Shuxing Chen, Editors, Geometry and Nonlinear Partial Differential Equations, 2002
 - 28 Valentin Afraimovich and Sze-Bi Hsu, Lectures on Chaotic dynamical Systems, 2002
 - 27 M. Ram Murty, Introduction to p-adic Analytic Number Theory, 2002
 - 26 Raymond Chan, Yue-Kuen Kwok, David Yao, and Qiang Zhang, Editors, Applied Probability, 2002
 - 25 Donggao Deng, Daren Huang, Rong-Qing Jia, Wei Lin, and Jian Zhong Wong, Editors, Wavelet Analysis and Applications, 2002
 - 24 Jane Gilman, William W. Menasco, and Xiao-Song Lin, Editors, Knots, Braids, and Mapping Class Groups—Papers Dedicated to Joan S. Birman, 2001
 - 23 Cumrun Vafa and S.-T. Yau, Editors, Winter School on Mirror Symmetry, Vector Bundles and Lagrangian Submanifolds, 2001
 - 22 Carlos Berenstein, Der-Chen Chang, and Jingzhi Tie, Laguerre Calculus and Its Applications on the Heisenberg Group, 2001
 - 21 Jürgen Jost, Bosonic Strings: A Mathematical Treatment, 2001
 - 20 Lo Yang and S.-T. Yau, Editors, First International Congress of Chinese Mathematicians, 2001
 - 19 So-Chin Chen and Mei-Chi Shaw, Partial Differential Equations in Several Complex Variables, 2001
 - 18 Fangyang Zheng, Complex Differential Geometry, 2000
 - 17 Lei Guo and Stephen S.-T. Yau, Editors, Lectures on Systems, Control, and Information, 2000

- 16 Rudi Weikard and Gilbert Weinstein, Editors, Differential Equations and Mathematical Physics, 2000
- 15 Ling Hsiao and Zhouping Xin, Editors, Some Current Topics on Nonlinear Conservation Laws, 2000
- 14 Jun-ichi Igusa, An Introduction to the Theory of Local Zeta Functions, 2000
- 13 Vasilios Alexiades and George Siopsis, Editors, Trends in Mathematical Physics, 1999
- 12 Sheng Gong, The Bieberbach Conjecture, 1999
- 11 Shinichi Mochizuki, Foundations of p-adic Teichmüller Theory, 1999
- 10 Duong H. Phong, Luc Vinet, and Shing-Tung Yau, Editors, Mirror Symmetry III, 1999
- 9 Shing-Tung Yau, Editor, Mirror Symmetry I, 1998
- 8 Jürgen Jost, Wilfrid Kendall, Umberto Mosco, Michael Röckner, and Karl-Theodor Sturm, New Directions in Dirichlet Forms, 1998
- 7 D. A. Buell and J. T. Teitelbaum, Editors, Computational Perspectives on Number Theory, 1998
- 6 Harold Levine, Partial Differential Equations, 1997
- 5 Qi-keng Lu, Stephen S.-T. Yau, and Anatoly Libgober, Editors, Singularities and Complex Geometry, 1997
- 4 Vyjayanthi Chari and Ivan B. Penkov, Editors, Modular Interfaces: Modular Lie Algebras, Quantum Groups, and Lie Superalgebras, 1997
- 3 Xia-Xi Ding and Tai-Ping Liu, Editors, Nonlinear Evolutionary Partial Differential Equations, 1997
- 2.2 William H. Kazez, Editor, Geometric Topology, 1997
- 2.1 William H. Kazez, Editor, Geometric Topology, 1997
 - 1 B. Greene and S.-T. Yau, Editors, Mirror Symmetry II, 1997

The heat kernel has long been an essential tool in both classical and modern mathematics but has become especially important in geometric analysis as a result of major innovations beginning in the 1970s. The methods based on heat kernels have been used in areas as diverse as analysis, geometry, and probability, as well as in physics. This book is a comprehensive introduction to heat kernel techniques in the setting of Riemannian manifolds, which inevitably involves analysis of the Laplace–Beltrami operator and the associated heat equation.

American Mathematical Society www.ams.org

International Press www.intlpress.com The first ten chapters cover the foundations of the subject, while later chapters deal with more advanced results involving the heat kernel in a variety of settings. The exposition starts with an elementary introduction to Riemannian geometry, proceeds with a thorough study of the spectraltheoretic, Markovian, and smoothness properties of the Laplace and heat equations on Riemannian manifolds, and concludes with Gaussian estimates of heat kernels.

Grigor'yan has written this book with the student in mind, in particular by including over 400 exercises. The text will serve as a bridge between basic results and current research.



For additional information and updates on this book, visit www.ams.org/bookpages/amsip-47